ഹ



# Geochemistry, Geophysics, Geosystems<sup>°</sup>

## **RESEARCH ARTICLE**

10.1029/2021GC009967

## **Special Section:**

Tethyan dynamics: from rifting to collision

#### **Key Points:**

- Unsteady kinematics of the India-Asia collision is a critical factor for the transition from N-S contraction to E-W extension in Tibet
- The strong Tarim basin produced a topographic slope in Tibet, which directed the eastward gravity flow at the time of slow convergence
- The eastward flow shows Poiseuille behavior, determining the sinistral and dextral shear in northern and southern Tibet, respectively

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

Correspondence to:

N. Mandal, nibir.mandal@jadavpuruniversity.in

#### Citation:

Maiti, G., Roy, A., Sen, J., & Mandal, N. (2021). Impact of decelerating India-Asia convergence on the crustal flow kinematics in Tibet: An insight from scaled laboratory modeling. *Geochemistry, Geophysics, Geosystems,* 22, e2021GC009967. https://doi. org/10.1029/2021GC009967

Received 10 JUN 2021 Accepted 29 OCT 2021

#### **Author Contributions:**

Conceptualization: Giridas Maiti, Nibir Mandal Data curation: Giridas Maiti, Arnab Roy, Joyjeet Sen Formal analysis: Giridas Maiti, Arnab Roy Funding acquisition: Nibir Mandal Investigation: Giridas Maiti, Arnab Roy, Joyjeet Sen

© 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

## Impact of Decelerating India-Asia Convergence on the Crustal Flow Kinematics in Tibet: An Insight From Scaled Laboratory Modeling

Giridas Maiti<sup>1</sup>, Arnab Roy<sup>1</sup>, Joyjeet Sen<sup>1</sup>, and Nibir Mandal<sup>1</sup>

<sup>1</sup>Department of Geological Sciences, Jadavpur University, Kolkata, India

**Abstract** The factors controlling the spatiotemporally varying deformation patterns in Tibet, a prolonged period (~50 to 19 ± 3 Ma) of NNE-SSW shortening, accompanied by eastward flow and orogenparallel extension in a later stage ( $19 \pm 3$  Ma to present-day), are still poorly constrained. Using viscous models, we performed scaled laboratory experiments with steady and unsteady state collision kinematics to address this issue. Our model Tibet under steady-state collision, irrespective of high (5.5 cm/yr) or low (3.5 cm/yr) indentation rates fails to produce the present-day crustal velocity fields and the deformation patterns, reported from GPS observations. An unsteady-state collision with decelerating convergence rates (5.5–3.5 cm/yr) is found to be a necessary condition for the initiation of eastward flow and ESE-WNW extensional deformations. The model results also suggest that the mechanical resistance offered by the rigid Tarim block resulted in crustal uplift at faster rates in western Tibet, setting a west to east topographic gradient, existing till present-day. This topographic gradient eventually polarized the gravitycontrolled flow in the east direction when the convergence velocity decelerated to ~3.5 cm/yr at around  $19 \pm 3$  Ma. Our model shows the present-day eastward flow in central Tibet follows nearly a Poiseuille type velocity profile, bounded by the Himalaya in the south and the Tarim basin in northern Tibet. This flow kinematics allows us to explain the preferential locations of crustal-scale dextral and sinistral faults in southern and northern Tibet, respectively. Finally, the present-day model crustal-flow velocity, strainrates, and topographic variations are validated with GPS and geological field data.

**Plain Language Summary** The Himalaya-Tibet Mountain system is believed to have formed due to crustal shortening and thickening in the course of India-Asia collision over the last 50 million years (Ma). The geological records, however, reveal a complex tectonic history of Tibet, showing dominantly north-south contractional deformations in the first phase (50 to 19 Ma), but replaced by eastward crustal flows and east-west extensional deformations after 19 Ma. The possible factors behind such tectonic transition in Tibet are not well known. This gap motivates us to explore them with the help of scaled laboratory modeling. Our model experiments allow us to rule out the possibility of steady state collision, either at fast or slow rates in setting the present-day flow patterns in the Tibetan plateau. We demonstrate decelerating India-Asia convergence velocity, 15 cm/yr at 50 Ma to 3.5 cm/yr at present-day is the key factor for the contractional to extensional tectonic transition in Tibet. The strong crustal block (Tarim basin) in northern Tibet has played a complementary role in forming a west to east topographic gradient that prompted the eastward gravity-driven crustal flows, as observed at present-day, when the convergence velocity slowed down to 3.5 cm/yr at 19 Ma.

## 1. Introduction

How the Tibetan plateau underwent deformations in space and time during the India-Asia collision has remained a subject of long debate (Bischoff & Flesch, 2018; Cook & Royden, 2008; Copley & Mckenzie, 2007; Houseman & England, 1993; Molnar & Tapponnier, 1975; Pusok & Kaus, 2015; Royden et al., 1997; Schellart et al., 2019; Tapponnier et al., 2001). A synthesis of the present-day GPS velocities (Figure 1a, Gan et al., 2007; Liang et al., 2013; Q. Wang et al., 2001; Zhang et al., 2004; Zheng et al., 2017) suggests that the crustal materials in the elevated plateau are currently undergoing an eastward "glacier-like" flow (21.6  $\pm$  2.5 mm/yr), with increasing speed towards the east (Figures 1b and 1c), and the flow eventually takes a clockwise turn around Eastern Himalayan Syntaxis (EHS) to channelize south-eastward, forming a fan-like front in the easternmost flank (Figure 1a) (Gan et al., 2007; Zhang et al., 2004). Present-day geodetic



Methodology: Giridas Maiti, Arnab Roy, Joyjeet Sen Project Administration: Nibir Mandal Resources: Nibir Mandal Software: Arnab Roy Supervision: Nibir Mandal Visualization: Giridas Maiti, Arnab Roy, Joyjeet Sen Writing – original draft: Giridas Maiti Writing – review & editing: Giridas Maiti, Arnab Roy, Nibir Mandal strain rates also suggest a dominant ESE-WNW extensional deformation to localize preferentially in southern and central Tibet (Zheng et al., 2017).

The deformation and crustal-flow kinematic history of Tibet, reconstructed from various geological records, however, indicate a strikingly different tectonic scenario in the past. Since the beginning of collision at  $\sim$ 50 Ma to  $\sim$ 19  $\pm$  3 Ma it was dominated by NNE-SSW directed contractional deformations (thrust faulting) both in its interior and southern margins (Y. Li, Wang, et al., 2015; Wang et al., 2014). During this time period the plateau had most of its crustal thickening to build the present-day raised topography with an average elevation of 5 km by 26 Ma (DeCelles et al., 2007; Fang et al., 2020). It is still ambiguous whether the topographic uplift took place uniformly or varied in space and time during this period. Several workers have proposed a non-coeval uplift hypothesis to explain the growth of the Tibetan plateau along east-west as well as north-south directions (Chung et al., 1998; Tapponnier et al., 2001; Wang et al., 2008). According to the outward growth model (Wang et al., 2008), the plateau grew southward and northward from an initially elevated proto-Tibetan plateau (presently Lhasa and Qiangtang blocks), which had already attained a topographic elevation of 3-4 km by ~40 Ma (Kapp et al., 2005; Murphy et al., 1997). On the other hand, compilation of thermochronological data along N-S transect of the plateau (Y. Li, Wang, et al., 2015) suggests, although central Tibet might have attained its elevated topography to some extent before 40 Ma, the plateau underwent an overall northward progressive contraction and uplift after the onset of India-Asia collision. The most striking event, the plateau witnessed during the collision, is marked by a dramatic transformation in its tectonic deformation pattern at around  $19 \pm 3$  Ma. At this stage ESE-WNW extension was onset in central and southern Tibet (Blisniuk et al., 2001; Lee et al., 2011; Kali et al., 2010; Styron et al., 2013; Y. Li, Wang, et al., 2015; McCallister et al., 2014; Wang et al., 2014) with the ongoing NE-SW contraction in the orogen margins. This tectonic transition is evident from initiation of a series of N-S trending high-angle normal faults or rifts in central and southern Tibet (Cooper et al., 2015; Ge et al., 2015). The extensional tectonics is shown to have continued till the present-day with a west to east younging trend in rift initiation (Bian, Gong, Zuza, et al., 2020; Cooper et al., 2015). During this period the contractional tectonics shifted its focus from the interior of the Tibetan plateau to its margins (S. Li, Currie, et al., 2015; Maiti & Mandal, 2021), including the regions of the Himalayan wedge (Main Central Thrust, 23-13 Ma; Main Boundary Thrust, 12-8 Ma; Main Frontal Thrust, 5-0 Ma) (Huyghe et al., 2001; Meigs et al., 1995; Montemagni et al., 2018; Najman, 2006; Soucy La Roche et al., 2018). On the other side, the north-eastern (Northern Qilian Shan thrust, 9-0 Ma) and eastern (Longmen Shan thrust, 10-0 Ma) margins of Tibet underwent fresh contractional deformation pulses at this late Cenozoic stage (Clark et al., 2005; E. Wang et al., 2012; Xu & Kamp, 2000; S. Yang et al., 2007). The central plateau thrust belts are eventually not major contributors to the present-day Tibetan tectonics (Y. Li, Wang, et al., 2015), as reflected from the earthquake focal mechanism solutions (Copley et al., 2011). In fact, the earthquake records indicate contraction deformations (thrust activities) to localize preferentially in the southern, north-eastern, and eastern margins of the Tibetan plateau, leaving its interior dominated by normal and strike-slip faulting (Figure S1 in Supporting Information S1). The thrust activities in central and southern Tibet became remarkably weak, but not totally inactive. For example, some recent studies have reported active thrusts from Gangdese (Taylor et al., 2021) and Qimen Tagh regions (Wu et al., 2020, 2021; Yakovlev, 2015).

The tectonic transition, NNE-SSW shortening during the initial phase of collision ( $\sim$ 50 to 19 ± 3 Ma) to a dominant ESE-WNW extension at around 19 ± 3 to 0 Ma in central and southern Tibet, is widely accepted and strongly backed by extensive geological evidences, for example, fault kinematics and sedimentary basin formation (Blisniuk et al., 2001; Cooper et al., 2015; Ge et al., 2015; Mitsuishi et al., 2012; Styron et al., 2013). But, what kind of geodynamic models can best describe such spatiotemporal variations of crustal deformation in the Tibetan plateau has remained an open-ended issue. A variety of models have been proposed to explain different modes of the Cenozoic crustal deformation pattern in the plateau and their relationship to topographic upliftment. The major models are: (a) intracontinental deformation of the plateau by rigid block indentation, and lateral crustal extrusion along large scale strike-slip faults (Tapponnier et al., 1982), (b) underplating of Indian crust beneath the Tibetan plateau as far as northern Qiangtang block, leading to crustal thickening and rapid surface upliftment in the plateau (DeCelles et al., 2002; Powell, 1986), (c) distributed crustal thickening by continuous north-south shortening of the entire plateau (England & Houseman, 1986), (d) convective removal of the Tibetan mantle lithosphere and Miocene upliftment of the Tibetan plateau (Molnar et al., 1993), (e) lower crustal channel flow dynamics for low-relief in the interior





**Figure 1.** (a) Present-day GPS velocity field in the Tibetan plateau (Gan et al., 2007). (b) Topographic profiles along CD (east-west), EF (northwest-southeast), and GH (southwest-northeast) sections (profiles are marked with dotted lines in panel a), showing a general west-to-east topographic gradient in the presentday Tibetan plateau. (c) Orogen-transverse variation of the east-west (N110°E) GPS velocity components, measured along N-S profiles (marked in solid colors in panel a). (d) Variation of the India-Asia convergence velocity with time. Note that different stages (S1, S2, and S3) with corresponding indentation velocity (cm/ yr) and crustal shortening ( $\epsilon$ ) are for the Type 3 experiment (discussed later).

of the plateau and steep topography in its eastern and southern margin (Bird, 1991; Clark & Royden, 2000; Zhao & Morgan, 1987), and (f) an outward growth model from a pre-existing proto-Tibetan plateau (Murphy et al., 1997; C. Wang et al., 2008). Yin (2000) proposed a completely different geodynamic model to explain E-W extension in the late Cenozoic time. According to his model, slab rollback of the western pacific plate gave rise to a regional lithospheric extension in entire southeast Asia, leading to Tibetan and Baikal rifting, and Shanxi graben formation. Despite a strong divergence in the prevailing views, it is generally accepted, based on the spectacular records of Tibet's long-term evolution history, that the evolution of plateau topography had an intricate relationship with the continental scale deformation episodes in Tibetan crust (DeCelles et al., 2002; England & Houseman, 1989; Y. Li, Wang, et al., 2015). Recent tectonic models have shown that the lateral crustal heterogeneities, such as rigid Tarim and Sichuan basins (Figure 1a) in the Asian plate have strongly influenced the topographic uplift and the long-term crustal deformations in the Tibetan plateau (L. Chen et al., 2017; Pusok & Kaus, 2015; Y. Yang & Liu, 2013).

Paleomagnetic data and plate reconstructions suggest that the India-Asia collision slowed the convergence rates, from 15 cm/yr at 50 Ma to 3.5 cm/yr at the present-day (Figure 1d) (Copley et al., 2010; Molnar & Stock, 2009; Pusok & Stegman, 2020). Such a decelerating convergence motion had the potential to reduce the tectonic forcing upon the deforming Asian plate, resulting in the gravitational collapse of the elevated plateau in the course of collision (Maiti et al., 2020; Rey et al., 2001). Copley and McKenzie's theoretical



model (2007) predicts this gravitational collapse as the driving factor in setting the present-day extensional tectonics in Tibet. On the other hand, several workers suggested eastward gravitational spreading of the whole Tibetan lithosphere, confined between two rigid blocks (Indian craton in the south and the Tarim and Qaidam basins in the north) controlled the rifting and strike slip faulting in the Tibetan plateau (Bajolet et al., 2015; Fournier et al., 2004; Yin & Taylor, 2011). From laboratory models, Bajolet et al. (2013) showed that the symmetric lateral extrusion between two rigid colliding plates, coupled with a late stage gravity collapse of the thickened crust modulated Tibetan tectonics, including the syntaxis formation on the eastern and western flanks of Tibet. This model is in agreement with Fournier et al.,'s (2004) study, suggesting a free or weakly confined eastward boundary of Tibet as a necessary condition for eastward gravitational spreading of the Tibetan crust. However, these earlier numerical and analogue experiments mostly ignored the temporally varying India-Asia collisional kinematics in their models, setting the experimental runs at a constant boundary condition, either with an average convergence velocity of 5 cm/yr (Beaumont et al., 2001; Y. Yang & Liu, 2013) or an average indentation rate of ~3.5 cm/yr (Schellart et al., 2019) for the entire 50 Ma collision event. Furthermore, most of the earlier laboratory models did not account for lateral crustal heterogeneities, particularly rigid Tarim and Sichuan basins situated in the north-western and easternmost regions of Tibet, respectively. Numerical simulations of England and Houseman (1989) took into account the effects of decreasing collision force, expressed in terms of Argand number, but their study excludes any possible effects of the lateral crustal heterogeneities (Tarim and Sichuan basins) around the Tibetan plateau. Despite a remarkable progress of tectonic modeling in recent time, it is yet to address the following questions- did the temporally varying collision kinematics play a critical role in the tectonic evolution of Tibet, and if so, what were the complementary effects of the lateral crustal heterogeneities? These issues constitute the central theme of our present article.

From indentation model experiments, we hypothesize that the decelerating convergence motion in India-Asia collision and the presence of lateral crustal heterogeneities are the most crucial factors in switching the tectonic transition in the Tibetan plateau, an initial period (~50 to  $19 \pm 3$  Ma) of NNE-SSW crustal shortening followed by initiation of eastward crustal flow and ESE-WNW extensional tectonics ( $19 \pm 3$  to 0 Ma), as observed in its present-day setting. To test this hypothesis, we considered three kinematic states: (a) constant fast-rate collision, (b) constant slow-rate collision, and (c) decelerating collision rate in running the laboratory model experiments. Our model experiment with the decelerating collision kinematics reproduces the current eastward crustal flow pattern in central Tibet revealed from the GPS studies. The N-S profile of the present-day eastward flow components is shown to follow Poiseuille behavior, where the Himalaya in the south and the Tarim basin in the north act as the guiding walls. Finally, we compare our model results with the present-day surface deformation pattern, estimated from GPS data and geological records in the Tibetan plateau, and provide a new tectonic evolution model for the plateau and its surrounding regions.

#### 2. Modeling Approach

We adopted a classical indentation tectonic model to perform analogue experiments of the India-Asia collision in laboratory (Fournier et al., 2004; Schellart et al., 2019; Tapponnier et al., 1982). The experiments were designed to study the Tibetan plateau deformation in the last ~50 Ma as a function of the three collisional kinematic states discussed in the preceding section. We use the experimental results to constrain the decelerating India-Asia collision rate as a critical collision kinematics required for setting the eastward crustal flow in Tibet. Our laboratory models allow us to explain the crustal velocity pattern, topography, and strain fields in space and time.

#### 2.1. Model Setup

The continental lithosphere is often modeled as a fluid continuum of Newtonian or power-law rheology to describe its large-scale deformation behavior on million years' time scales (England & McKenzie, 1982). As the horizontal length scale of the continuum system (Tibetan plateau) far exceeds the lithospheric thickness, several workers have developed their tectonic models within a framework of thin viscous sheet (TVS) approximation (Bischoff & Flesch, 2018; Houseman & England, 1986; Yang & Liu, 2013). The TVS approximation (England & McKenzie, 1982) simplifies depth-wise rheological variations in the lithosphere by





**Figure 2.** (a) A schematic diagram of the laboratory model setup (top view) used for India-Asia indentation experiment. Deformation in Asia has been imposed by NNE-directed movement of the Indian block, attached to a step motor. (b) A schematic presentation of the vertical section XY (marked in panel a) of the model setup (vertical axis exaggerated).

considering a vertically averaged and approximated lithospheric rheology, and treats the system as a single continuum medium. We chose a two-layer continuum model to perform scaled laboratory experiments on Tibet.

In our model, the continental deformation in the Asian domain has been enforced by imposing external velocities in the Indian indenter, attached to a step motor (Figure 2a). It is noteworthy that the dynamics of convergent margins involves a balance of the far-field tectonic forces, orogenic loading and other forces (Clark, 2012; Iaffaldano et al., 2006; Regard et al., 2005), which eventually determines the kinematics in orogens. However, to investigate exclusively the effects of reducing convergence rate on the mode of Tibetan crustal deformations, we chose a kinematic model boundary condition by externally controlling the velocities at the model boundaries. This kinematic approach to modeling is similar to those used in many earlier India-Asia collision models (Bajolet et al., 2015; Bischoff & Flesch, 2018; Cook & Royden, 2008; Fournier et al., 2004; Maiti & Mandal, 2021; Schellart et al., 2019; Y. Yang & Liu, 2013). We simulate approximately north-east ward Indian indentation into Asia. The northern boundary (leading) of the Indian indenter represents the contact between the Indian plate and the Himalaya-Tibetan plateau. A recent analysis of the seismic tomography and earthquake locations data suggests that the underthrusting of the Indian plate beneath the Tibetan plateau reaches up to  $\sim$ 300–500 km north of the Himalayan front (Craig et al., 2020). To include this subduction induced simple shear beneath the Asian plate, we have designed the Indian indenter, allowing a part of it to underthrust for about 350 km north of the mountain front (Figure 2b).

Unlike previous laboratory experiments of India-Asia collision (Bajolet et al., 2015; Fournier et al., 2004; Schellart et al., 2019; Tapponnier et al., 1982), we have incorporated several important additional features in our present model. One of them replicates the presence of lateral crustal heterogeneities that is, the inclusion of rigid Tarim and Sichuan basins, situated in the northern and the eastern margins of the Tibetan plateau, respectively. They act as a region of strong lithosphere (Clark et al., 2005; Dewey et al., 1988; Penney & Copley, 2021), as predicted from observations of fast seismic velocities down to a depth of 200 km (Li et al., 2006), measurements of low heat flow rate (Wang, 2001), and present-day GPS velocity data (Gan et al., 2007; Liang et al., 2013; Zhang et al., 2004). On the other hand, we have considered an arcuate plate boundary of the Indian plate. It is noteworthy that all the previous analogue India-Asia collision models used the Indian indenter ideally rectangular in shape, forming a straight plate contact. But, the present-day southern margin of the Tibetan plateau, that is, India-Asia contact, is strongly curved. The plate reconstructions and rotations measured from palaeomagnetic data (as summarized in Van Hinsbergen et al., 2011) suggest that this arcuate geometry existed since the onset of continent-continent collision in this region. Copley (2012) has shown how this curved plate contact influenced the gravitational spreading of the Tibetan plateau in the late stage of the orogeny. As our present study primarily aims to investigate the entire tectonic evolution of the plateau, including its late-stage gravity collapse, we choose such a curved plate contact to model the India-Asia collision system.

There are a number of advantages in analogue modeling used in this study. (a) Analogue models provide us an excellent scope for studying the four-dimensional (three-dimensional space + time) evolution of Tibet. (b) They have a free top surface, allowing the continuum to deform and produce mountain topography and plateau morphology, which can be easily compared with the natural prototype that is, the Himalaya-Tibet Mountain system. (c) The scaled analogue models reproduce large scale (i.e., 3,500 km × 5,000 km continental lithosphere) crustal flows, which can be studied in high resolution to compare the model results with natural data.

#### 2.2. Material Properties and Scaling

We have scaled our analogue model to the natural prototype setting, keeping geometric similarity of the model domain, similarity in the temporal variation of collision kinematics, and downscaling of the crustal and mantle rheology (Hubbert, 1937; Ribe & Davaille, 2013; Schellart & Strak, 2016). To model the India-Asia collision system we have used a length scale factor (nature to model ratio):  $l_r = 1.25 \times 10^{-7}$ . This scaling yields a model dimension of 40 cm to represent an approximately 3,200 km east-west length of the Tibetan plateau. Similarly, a 40 km thick crustal lithosphere in the initial model is represented as 0.5 cm thick viscous layer. The model crustal lithosphere rests over an initially 2 cm thick layer, which represents a sub-crustal mantle region. The whole setup is contained within an experimental tank. At the contact between the model base and the basal plate of the experimental tank, we have used liquid soap (viscosity ~0.5 Pa s) to minimize friction to the model. To ensure the lubrication at the desired level, we performed independent experiments and evaluated base normal velocity profiles, calculated from particle image velocimetry (PIV) on the model side faces (details provided in Figure S2 in Supporting Information S1).

Model materials with different rheological properties were chosen to appropriately scale the crustal lithosphere and underlying sub-crustal mantle region in the experiments. A glucose syrup (syrup-1) of constant density ( $\rho_{cl} = 1,406 \pm 3 \text{ kg/m}^3$ ) and viscosity ( $\mu_{cl} = 305 \pm 5 \text{ Pa s}$ ) is used to model the whole (40 km thick) crustal-lithosphere as a single continuum, representing its depth-averaged physical properties in the corresponding natural prototype (viscosity:  $3 \times 10^{21}$  Pa s; density: 2,780 kg/m<sup>3</sup>) (Jiménez-Munt et al., 2008). The model to nature scaling factors for viscosity ( $\mu_{z}$ ) and density ( $\rho_{z}$ ) are found to be 10<sup>-19</sup> and ~0.5, respectively. We have used another variety of glucose syrup (syrup-2) with different density and viscosity  $(\mu_{sm} = 28 \pm 5 \text{ Pa s}; \rho_{sm} = 1,510 \pm 5 \text{ kg/m}^3)$  to simulate the sub-crustal mantle region (viscosity: ~10<sup>20</sup> Pa s; density: 3,200 kg/m<sup>3</sup>) (Jiménez-Munt et al., 2008). To obtain the right density of glucose syrup-2, we have mixed water soluble sodium carbonate (Na<sub>2</sub>CO<sub>3</sub>. 10 H<sub>2</sub>O,  $\rho = 2.53$  gm/cm<sup>3</sup>) with a low-viscosity glucose syrup ( $\mu = 10 \pm 3$  Pa s and  $\rho = 1,392 \pm 5$  kg/m<sup>3</sup>). The lower  $\mu_{sm}$  and higher  $\rho_{sm}$  of syrup-2 than that of syrup-1 allows the base of the crustal lithosphere to deform vertically and maintain an isostatic equilibrium of the model topography during the experimental run. We have chosen commercial plasticine mixed with oil (viscosity =  $3 \times 10^5$  Pa s; density = 1.5 gm/cc) (Zulauf & Zulauf, 2004), which is relatively much stronger than the glucose syrup-1, to represent rigid Tarim and Sichuan basins in the model. The viscosity ratio in the model is then in the order of  $\sim 10^3$ , which agrees well with the numerical estimates of the viscosity ratio (~10<sup>2</sup> to 10<sup>3</sup>) between rigid Tarim basin and Tibetan upper crust (Clark & Royden, 2000; Flesch et al., 2001; Penney & Copley, 2021).



#### Table 1

List of Model Parameter Values Used for Laboratory Experiments, Along With Their Corresponding Values for the Natural Prototypes (Cook & Royden, 2008; Jiménez-Munt et al., 2008; Liu & Yang, 2003; Schellart et al., 2019)

Parameters	Model value ( <i>m</i> )	Nature value ( <i>n</i> )	Ratio ( <i>m/n</i> )
Gravity(g)	9.81 m/s <sup>2</sup>	9.81 m/s <sup>2</sup>	$g_r = 1$
Length ( <i>l</i> )	40 cm	3,200 km	$l_r=1.25\times 10^{-7}$
Density of crustal lithosphere ( $\rho_{cl}$ )	$1,406 \pm 3 \text{ kg/m}^3$	2,780 kg/m <sup>3</sup>	$\rho_r = 0.47 \text{ to } 0.5$
Density of sub-crustal mantle ( $\rho_{sm}$ )	$1,510 \pm 5 \text{ kg/m}^3$	3,200 kg/m <sup>3</sup>	
Viscosity of crustal lithosphere $(\mu_{cl})$	315 ± 5 Pa s	$3 \times 10^{21}$ Pa s	$\mu_r = 1 \times 10^{-19}$
Viscosity of sub-crustal mantle $(\mu_{sm})$	28 ± 3 Pa s	$3 \times 10^{20}$ Pa s	
Time ratio $(t_r)$	~50 s	1 Ma	$t_{\rm r} = 1.6 \times 10^{-12}$
Velocity ratio $(U_r)$	$U_m$	$U_n$	$U_r = 7.813 \times 10^4$

*Note.* The model scaling factors are provided in the extreme right column.

Considering an extremely slow motion in the geodynamic setting, the dynamic scaling must satisfy the following relation (Ghosh et al., 2020; Marques & Mandal, 2016),

$$\mu_r = \rho_r \times l_r \times t_r \tag{1}$$

Equation 1 yields a time ratio:  $t_r = 1.6 \times 10^{-12}$ . That means, 1 Ma time in nature equals to ~50 s in the scaled model. Now, using the relation:

$$U_r = l_r / t_r, \tag{2}$$

the velocity ratio in our model is:  $U_r = 7.813 \times 10^4$ . According to this scaling factor, Indian plate's indentation velocity of 5.5, 4.5, and 3.5 cm/yr in nature equates to 8.17, 6.69, and 5.20 mm/min in the model, respectively. The total model runtime in our experiments was ~42 min, which is equivalent to ~50 Ma in nature. All the scaling parameters are listed in Table 1.

#### 2.3. Recording of Experiments

For a quantitative analysis of the model crustal flow velocity and topography development in each experiment, we used a particle image velocimetry (PIV) system, which consisted of four synchronized cameras systematically aligned around the experimental box: three at the top and one at the side of the model set-up (Figure S3 in Supporting Information S1). The top cameras were placed  $\sim$ 80 cm above the set-up table, with 10°–15° angles between them to capture sharp images (Galland et al., 2016). All four cameras were standard Digital single-lens reflex cameras (Nikon D3400 24.2 MP) with a shutter speed of 1/125 and a focal length of 24 mm. The operation of these four cameras was synchronized with the help of open-source software, digiCamControl in a single computer system.

We used the *structure-from-motion* technique to retrieve quantitative surface deformation in the experiments (Galland et al., 2016). A stand-alone spatial photogrammetry software Micmac (Galland et al., 2016; Girod, 2012) was used to process the photographs and create a series of digital elevation models (DEMs) of the evolving topographic surface. The high-resolution DEMs were constructed after defining ground control points and the scale between the model and the images (Galland et al., 2016). The DEMs from the successive snaps of a model run gave a detailed time-series configuration of the evolving surface topography. We adopted this approach for a quantitative analysis of the topographic evolution in the Tibetan Plateau.

The method of high-resolution digital image correlation was implemented to track and analyze the evolving displacement, stress and strain fields in our scaled analogue models. Some raw photographic images of our model experiments are provided as examples in supporting information (Figure S4 in Supporting Information S1). We used PIVLab (Thielicke & Stamhuis, 2010), an open-source image correlation software running in MATLAB, to cross-correlate temporally successive orthorectified image pairs, captured with the top-view cameras and processed with Micmac (Rupnik et al., 2017). Unlike normal photographs, orthorectified



images are adjusted for topographic relief, lens distortion and camera tilt and they provide an accurate representation of the photographed surface. The stress and strain fields were deduced with high accuracy. We performed several such quantitative analyses of the model observations to study the effect of the India-Asia convergence kinematics on the Tibetan Plateau.

## 3. Reference Experiments

Plate reconstructions studies suggest that at the time of India-Asia collision, ~50 Ma ago, the Indian subcontinent was located at ~2,500 km south of its current position (Johnson, 2002; Molnar et al., 1993; Rowley, 1998; Royden et al., 2008; Van Hinsbergen et al., 2011). The amount of shortening involved in the evolution of the Himalaya-Tibet system is, however, a complicated issue. Again, there is a serious uncertainty with the partitioning of total shortening into the Himalayan range and Asia as the available estimates of Asian shortening vary on a wide spectrum (Ingalls et al., 2016; Johnson, 2002; Van Hinsbergen et al., 2011; Yakovlev, 2015). Recently, Schellart et al. (2019) demonstrated from experimental studies that  $1,740 \pm 300$  km of India–Eurasia convergence since the beginning of collision has been accommodated by indentation driven Central, East and Southeast Asian deformation. Their estimate agrees with earlier data (Huang et al., 2015). Based on Schellart et al.'s (2019) estimates, we have considered an indentation movement of 1,800-2,000 km in our model experiments. This India-Asia convergence requires an indentation rate, that is, the advance velocity of the leading edge of the Indian plate,  $\sim$ 3.5 cm/yr, when averaged over  $\sim$ 50 Ma (Schellart et al., 2019). The geological records, however, clearly suggest that the India-Asia collisional kinematics never remained steady, but decelerated with time (Copley et al., 2010; Molnar & Stock, 2009; Pusok & Stegman, 2020). We thus aimed to explore from our laboratory model experiments the possible effects of such decreasing indentation velocity on the crustal flow and deformation patterns in Tibet. To recognize this effect, we performed three types of experiments, and compared their results. Type 1 experiment simulated a steady fast-rate collisional kinematics, setting the indentation rate ( $I_{\text{FAST}}$ ) at a constant value of 5.5 cm/yr. Type 2 experiment simulated a steady, but slow-rate collisional kinematics with a constant indentation rate ( $I_{SLOW}$ ) of 3.5 cm/yr, which represents the average indentation rate. In the Type 3 experiment, we generated an unsteady collisional kinematics, beginning with a fast indentation rate of 5.5 cm/yr, but decelerating ( $I_{\text{DEC}}$ ) to 3.5 cm/yr with time. This experiment exactly represents the reducing India-Asia convergence velocity reported in the literature (Copley et al., 2010; Molnar & Stock, 2009). In all the three types of experiments, the Indian indenter's leading edge was positioned at a distance around 1,800-2,000 km south from its present-day position. The total indentation of 1,800-2,000 km in the model is expressed in percentage as  $\epsilon = 100\%$ .

#### 3.1. Type 1 Model Experiment: Steady Fast-Rate Collision

In this experiment of steady fast-rate ( $I_{FAST} = 5.5$ . cm/yr) collision at  $\epsilon = 30\%$  convergence the model produced a NNE directed high velocity (5 cm/yr) crustal flow field in southern Tibet, gradually weakening nearly to 0 cm/yr in the northern margin of Tibet (Figure 3a-i). When  $\epsilon = 50\%$ , the NNE directed flow field grew northward, setting strong flow velocities (~4 cm/yr) in the central Tibetan region. But, the northern-most part of the plateau still remained nearly undisturbed (Figure 3a-ii). At  $\epsilon = 70\%$ , the northward crustal flow interacted with the rigid Tarim basin situated in the north, and started to deflect towards north-eastern Tibet. The crustal flow velocities also showed a strong spatial heterogeneity, ~5 cm/yr dominantly in southern and central Tibet, whereas ~3 cm/yr in northern Tibet (Figure 3a-ii). This velocity gradient clearly suggests NNE-SSW contractional deformations in a large part of Tibet. At  $\epsilon = 100\%$ , that is, after the completion of total ~2,000 km Indian indentation to the present-day condition, the entire Tibetan plateau continued to evolve with dominant NE-SW contractional deformations, maintaining crustal flow velocities nearly 5 cm/yr in southern Tibet, ~3.5 cm/yr in central Tibet, and ~2.5 cm/yr in northern and north-eastern Tibet (Figure 3a-iv). During the entire stage of collisional event, the model did not produce any clockwise rotation of the crustal flows around EHS, as observed in the present-day Tibet (Figure 1a).

We plotted the principal strain axes on the second invariant of strain-rate tensor maps to study the spatio-temporal variations of crustal deformation in Type 1 model experiments. The plots show intense deformation (high strain rates  $\sim 4.5 \times 10^{-8} \text{ yr}^{-1}$ ) localization preferentially near the leading edge of the Indian indenter in the initial stage ( $\epsilon = 30\%$ ) of India-Asia collision (Figure 3b-i). With progressive collision the high





**Figure 3.** Top views of Type 1 model experiment with constant high-indentation velocity ( $I_{FAST} = 5.5 \text{ cm/yr}$ ). (a) Snapshots of the crustal flow patterns taken at progressive stages of the convergence ( $\epsilon$ ). (b) Corresponding plots of the second invariant and the principal axes of horizontal strain-rate tensor (blue is compression, red is extension). (c) Topographic profiles along AB (North-South) and CD (West-East) sections in the model for different percentages ( $\epsilon$ ) of the convergence. AB and CD sections are marked in panel a-i.

strain-rate zone propagated northward. During  $\epsilon = 30\%-70\%$  convergence the northern and the north-eastern regions of Tibet had crustal shortening at strain rates of 6 to  $2.0 \times 10^{-8}$  yr<sup>-1</sup> (Figures 3b-ii, iii, iv). The principal strain axes plot suggests that the entire collisional event underwent contractional deformations broadly in the NNE-SSW direction.

The topographic analysis reveals that at 30% convergence, the plateau developed a northward topographic slope of 0.02°, suggesting a higher uplift rate in southern Tibet, compared to the northern region. At this stage the plateau gained virtually no topographic slope (0.01°) in the east-west direction (Figure 3c-i). For  $\epsilon = 70\%$ , the northward slope steepened further to 0.1°, but the E-W topographic slope remains almost steady at 0.02° (Figure 3c-ii). Increasing convergence resulted in a topographic modification, marked by the increase of the northward topographic slope from 0.1° to 0.6° when  $\epsilon = 100\%$  (Figure 3c-iii).

#### 3.2. Type 2 Model Experiment: Steady Slow-Rate Collision

The Type 2 experiment was similarly run at a steady, but relatively slow rate ( $I_{\text{SLOW}} = 3.5 \text{ cm/yr}$ ) to cover the total ~2,000 km convergence. The model run at  $\epsilon = 32\%$  produced high crustal flow velocities (3.5 cm/ yr) in the southern Tibetan region, where the flow had broadly a NNE-ward trend. However, central and northern Tibet were left virtually undisturbed, showing little or no crustal flow (Figure 4a-i). At  $\epsilon = 50\%$  the high-velocity (~3 cm/yr) crustal flow regime in the southern region propagated northward to set a strong crustal flow (~2 cm/yr) in central Tibet (Figure 4a-ii). The northern part of Tibet also became kinematically





**Figure 4.** Top views of Type 2 model experiment with constant and low indentation velocity ( $I_{SLOW} = 3.5 \text{ cm/yr}$ ). (a) Crustal flow patterns for increasing convergence ( $\varepsilon$ ). (b) Corresponding second invariant and principal axes of the horizontal strain-rate tensor (blue is compression, red is extension) in the model. (c) Topographic profiles along AB (North-South) and CD (West-East) sections for different percentages of convergence ( $\varepsilon$ ). AB and CD sections are marked in panel a-i.

active, but with significantly low velocities (~1 cm/yr). After  $\epsilon = 70\%$  strong high-velocity crustal flows still dominated in southern (~3 cm/yr) and central Tibet (~2 cm/yr), and at the same time the flows in northern Tibet - strengthened by increasing their velocity to ~1.4 cm/yr (Figure 4a-iii). The rigid Tarim basin deflected crustal-flows towards north-eastward, but at much lower magnitudes, as compared to that observed in the Type 1 experiment. At the completion of experimental run (i.e.,  $\epsilon = 100\%$ ) the model maintained the same kinematic state of crustal flows with slightly reduced magnitudes (~3 cm/yr in southern Tibet and ~1.8 cm/yr in northern Tibet; Figure 4a-iv). This Type-2 experiment also failed to produce the clockwise turning crustal-flow field around EHS of the present-day Tibet (Figure 1a).

The second invariant mapping of strain-rate tensor from Type 2 model experiments reveals that intense contractional deformations localized at the rate of  $2.0 \times 10^{-8}$  yr<sup>-1</sup> in front of the Indian indenter during the initial stage of convergence ( $\epsilon = 32\%$ ) (Figure 4b-i). In course of the convergence motion ( $\epsilon = 30\%$ -70%) the deformation front propagated northward, inducing grossly NE-SW contraction at  $2.4 \times 10^{-8}$  yr<sup>-1</sup> in central Tibet (Figures 4b-ii, iii). At  $\epsilon = 100\%$ , high strain rates (1.2 to  $2.4 \times 10^{-8}$  yr<sup>-1</sup>) localized preferentially in central and north-western Tibet, leaving the northern and north-eastern regions of Tibet virtually undeformed (Figure 4b-iv).

The model run at  $\epsilon = 30\%$  produced a gentle northward slope (0.02°), but negligibly small (0.009°) eastward slope (Figure 4c-i). With increasing convergence the northward topographic slope of Tibet steepened from 0.02° to 0.03° when  $\epsilon = 70\%$ . Meanwhile, the eastward topographic slope had a small value, ~0.01°





**Figure 5.** Top views of a Type 3 model run with decreasing indentation velocity ( $I_{DEC}$ ) from 5.5 to 3.5 cm/yr. (a) Snapshots of the crustal flow patterns taken at different percentage of the convergence ( $\epsilon$ ). (b) Changes in the second invariant and principal axes of horizontal strain-rate tensor (blue is compression, red is extension) with increasing  $\epsilon$ . (c) N-S and E-W topographic profiles, AB and CD sections for different  $\epsilon$ .

(Figure 4c-ii). On completion of the experimental run ( $\epsilon = 100\%$ ), the model Tibet attained a N-S slope of 0.05°, whereas the E-W slope increased to 0.02° (Figure 4c-iii).

To highlight the main findings from the two steady-state collision experiments (Type 1 and Type 2), irrespective of fast or slow convergence rates, the collision did not produce eastward crustal flows or clockwise rotation of the crustal flow field around EHS in model Tibet at any stage of the convergence event. None of them gave rise to the present-day E-W extensional strain in central Tibet, as revealed from GPS observations (discussed later in section). We, therefore, chose to run the unsteady collision experiments with decelerating convergence rates, as applicable for the India-Asia collision event (Copley et al., 2010; Molnar & Stock, 2009).

#### 3.3. Type 3 Model Experiment: Decelerating Collision

To perform this unsteady collision experiment, we reduced the convergence velocity  $(I_{DEC})$  stepwise in three stages, approximated to the decreasing India-Asia convergence velocity shown in Figure 1d. In stage S1 ( $\epsilon = 0\%$ -40%;  $I_{DEC} = 5.5$  cm/yr), the model produced a strongly heterogeneous NNE-directed crustal-flow regime at  $\epsilon = 30\%$ . The flow occurred at high velocities (~5 cm/yr) in southern Tibet, but decreased to almost zero at the northern boundary (Figure 5a-i). In stage S2 ( $\epsilon = 41\%$ -70%;  $I_{DEC} = 4.5$  cm/yr), the flow regime propagated northward to induce crustal flow velocity ~4.2 cm/yr in central plateau at  $\epsilon = 50\%$ . The model produced flows also in northern Tibet, although at much lower rates (~1.5 cm/yr) (Figure 5a-ii). At this stage the rigid Tarim basin acted as a mechanical barrier, and reduced the effective width of the western

Tibetan plateau to make it much narrower than its eastern flank. This west to east variation in the mechanical setting caused the crustal flows to take a north-east ward turn in the eastern parts of Tibet (Figure 5a-iii). In stage S3 ( $\epsilon = 71\%$ -100%;  $I_{\text{DEC}} = 3.5$  cm/yr) of collision, the model showed a remarkable change in the crustal flow pattern (Figure 5a-iv). Southern Tibet underwent NNE-directed flows with an average rate of 3.5 cm/yr, which reduced to ~2 cm/yr in north-eastern Tibet with an azimuthal change grossly to ENE. Second, the flow field developed a distinct region of east-directed crustal flows with magnitudes of 1.5–2 cm/yr in the central and southern parts of the plateau (Figure 5a-iv), which was not observed in the steady collision experiments. Interestingly, the eastward crustal flow, after facing resistance from the mechanically strong Sichuan basin in the east, took a clockwise turn to channelize through the passage in between the EHS and Sichuan basin, and escaped southward, forming a fan-like front. This region accelerated the crustal flows, with magnitudes reaching ~3 cm/yr (Figure 5a-iv). At the final stage ( $\epsilon = 100\%$ ), the model produced a crustal-flow pattern strikingly similar to the present-day GPS velocity field (Figure 1a; Liang et al., 2013; Zhang et al., 2004).

In the initial stage ( $\epsilon = 30\%$ ) of collision the Type 3 model showed localization of high strain rates ( $3.2 \times 10^{-8} \text{ yr}^{-1}$ ) in front of the leading Indian indenter edge (Figure 5b-i). With on-going collision ( $\epsilon = 30\%$ -70%) the deformation front propagated northward to increase the strain rates in central and NW Tibet at 3.2 to  $4 \times 10^{-8} \text{ yr}^{-1}$  (Figures 5b-ii, iii). At  $\epsilon = 100\%$ , southern, north-western, and north-eastern Tibet became active locations for deformations with high strain rates (2 to  $3.2 \times 10^{-8} \text{ yr}^{-1}$ ) (Figure 5b-iv). Interestingly, at this stage the region between EHS and Sichuan basins also started to deform at high strain rates ( $3.2 \times 10^{-8} \text{ yr}^{-1}$ ). On the other end, central Tibet was dominated by ESE-WNW extensional strains of 0.8 to  $2.2 \times 10^{-8} \text{ yr}^{-1}$  (Figure 5b-iv).

The decelerating collision kinematics led to characteristic topographic evolution in model Tibet. In stage S1 ( $\epsilon = 30\%$ ) the southern plateau gained higher topographic elevation to form a gentle northward slope of 0.03°, and almost an equal slope of 0.04° in the E-W direction (Figure 5c-i). Towards the end of stage S2 ( $\epsilon = 70\%$ ), the model plateau underwent a remarkable topographic modification, setting steep topographic slopes both in the northward (0.07°) and eastward (0.1°) directions (Figure 5c-ii). The eastward topographic gradient directed the gravitational collapse of the elevated plateau to facilitate crustal flows from western and central Tibet towards the south-eastern and north-eastern region. At the end of stage S3 ( $\epsilon = 100\%$ ), the topographic gradients reduced both along the N-S (0.02°) and E-W (0.04°) directions (Figure 5c-iii).

## 4. A Comparative Analysis of Type 1, 2 and 3 Experiments

All the three types of model experiments showed that the rigid Tarim basin at the northern margin of Tibet acted as a mechanical barrier to the northward crustal-flows induced by the Indian indenter. The rigid block forced the crustal flows to redirect their path in the northeast direction (Figures 3a, 4a, and 5a), and contributed to the development of relatively higher topography in western Tibet. The three types of experiments also suggest that the deformation fronts in Tibet progressively migrated northward after the initial phase of collision (Figures 3b, 4b, and 5b). However, the spatio-temporal variations of crustal flow patterns and topographic upliftment produced in the three models are markedly different from one another. Type 1 and Type 2 models maintained a constant northward to north-eastward crustal-flow azimuth, for the entire period of experimental run. They never set any dominant eastward flow at any stage of the convergence history. In contrast, the Type 3 model produced a distinct eastward flow field after 70% convergence when the indentation velocity ( $I_{DEC}$ ) reduced to 3.5 cm/yr (Figure 5a).

#### 4.1. Eastward Crustal-Flow Pattern

We will now compare the present-day (for  $\epsilon = 100\%$ ) crustal-flow patterns produced in the three types of model experiments with the present-day GPS velocity field (Figure 6). Type 1 model produces crustal-flows dominantly in the NNE direction, which continued to occur with high velocities in the extreme north-east corner, without any sign of clock-wise turn against the EHS (Figure 3a-iv). The flow pattern in eastern Tibet under this steady fast-rate collision condition markedly disagrees with the GPS velocity field (Figure 6a and 1a). Similarly, the model topography (steep northward slope, 0.6° and 0.04° E-W slope; Figure 3c-iii) does not match with the present-day topographic slopes of the Tibetan plateau (Figure 1b). Type 2 model



## Geochemistry, Geophysics, Geosystems



**Figure 6.** Comparison of the present-day GPS velocities (blue arrows) in the Tibetan plateau (Zhang et al., 2004) with the model predicted crustal-flow velocities (black arrows) obtained from the three types of model experiments: (a) Type 1 with  $I_{FAST} = 5.5 \text{ cm/yr}$ , (b) Type 2 with  $I_{SLOW} = 3.5 \text{ cm/yr}$ , and (c) Type 3 with  $I_{DEC}$  decreasing from 5.5 to 3.5 cm/yr. Note that the crustal-flow velocities in the Type 3 model match closely with the present-day GPS velocities.

also shows a discernible mismatch of the crustal-flow pattern with the GPS velocity field (Figure 6b). The high-velocity crustal flow regime occurs only in southern Tibet, but without much transmission of contractional deformation in north-eastern Tibet (Figure 4a-iv).

The crustal flow pattern produced in the Type 3 model shows an excellent agreement with the present-day GPS velocity field in Tibet (Figure 6c). The model topographic slopes along E-W and N-S also show a good match with the present-day topographic slopes (Figures 5c-iii, and 7b). Moreover, a time-series analysis of topographic uplift from the Type 3 model indicates that the rigid Tarim basin in the north acted as a mechanical obstruction to reduce the horizontal crustal flow velocity, and facilitated western Tibet to grow vertically faster at an average  $\sim 0.24$  mm/yr than the eastern Tibet at  $\sim 0.10$  mm/yr in the initial 25–30 Ma of the India-Asia collision. Such differential uplift rates resulted in a first-order eastward topographic slope during the phase ( $\sim$ 50–19 ± 3 Ma) of high indentation velocity ( $\geq$ 4.5 cm/yr) (Figure 7c). The topographic slope underwent gravity collapse in the phase of reduced indentation velocity to 3.5 cm/yr, which in turn set the eastward crustal flow in central Tibet. The gravitational collapse and onset of the eastward crustal flow can be theoretically interpreted in terms of the Argand number Ar, which is the ratio between two stresses, one arising from crustal thickness contrasts and the other required to deform the material at the ambient strain rates (England & McKenzie, 1982). According to this theoretical formulation, the balance between the gravity and the resistive viscous forces determines the rate of gravity-driven crustal flows in an orogen. A decrease in the convergence velocity results in an increase in Ar, that is, a greater influence of the gravitational potential energy (GPE) in the orogen dynamics. In the higher Ar condition gravity forces arising from the crustal thickness contrast thus captured the dynamics of Tibetan tectonics, setting in an eastward flow in the plateau when the convergence rate significantly dropped in Stage-S3. On the other end, the rigid Sichuan basin on the eastern flank of Tibet resisted the rapid eastward crustal-flow originated from



## Geochemistry, Geophysics, Geosystems



**Figure 7.** (a) Present-day DEM topography of the Tibetan plateau and locations of some active thrusts in Miocene time. Abbreviations are: NST, Nanshan Thrust belt; NQLT, Northern Qilian Shan thrust belt; LMST, Longmen Shan Thrust; GLT, Ganzi–Litang thrust belt; WKT, Western Kunlun Thrust (b) Comparison of the present-day model topographic profiles, obtained from the Type 3 model experiment ( $I_{DEC} = 5.5$  to 3.5 cm/yr) with the DEM profiles of the present-day plateau. (c) Calculated 3D topography from the Type 3 model produced after 70% of total convergence. Note that we have observed different rates (average) of topographic uplift from west to east in the Tibetan plateau.

the central plateau, and eventually forced it to take a clockwise turn around the EHS and flow towards the south with a fan-like front in the Burmese plate region (Figures 5a ii-iii). Topographic analysis suggests the Type 3 model developed a steep  $(0.1^{\circ})$  eastward slope after 70% convergence, which subsequently reduced to 0.04° at the present-day (Figure 5c). This clearly indicates that the eastward flow was driven by topography-controlled gravity current (Copley & Mckenzie, 2007), which occurred in the decelerating kinematic condition of Indian plate indentation. Earlier laboratory models for India-Asia collision also suggested that the gravitational collapse of the Tibetan plateau governed the late stage extensional tectonic in the orogen (Bajolet et al., 2013, 2015; Fournier et al., 2004). Fournier et al. (2004) showed that a confined western and a free or weakly confined eastern boundary of Tibet directed the gravitational spreading of Tibet in the east direction. Our model adds that the combined effect of lateral crustal heterogeneity (Tarim and Sichuan basins) and a reducing India-Asia convergence velocity controlled the differential topographic upliftment from west to east, and its subsequent gravitational collapse in the late stage (last ~18 Ma). This model also suggests that the rigid Sichuan basin in the eastern boundary of Tibet forced the eastward flow to take a turn in the south eastern direction, forming a fan like front in the Burmese platelet.

#### 4.2. Contractional to Extensional Strain Fields

The three types of experiments had remarkable differences in the spatio-temporal variations of their strainrate fields. In Type 1 and Type 2 models the contractional zones progressively migrated from southern to







**Figure 8.** Comparison of the present-day principal axes of horizontal strain-rate tensor (blue is compression, red is extension), derived from (a) GPS velocities (after Zhang et al., 2004) and (b) model results. (c) Plot of the present-day model shear-strain rate ( $\dot{\gamma}$ ), showing sinistral and dextral shear strain localization in the northern and southern boundaries of Tibet, respectively. Comparison between the model fields of shear sense and the present-day strike-slip faults in Tibet (fault motions are adopted from Tapponnier et al., 2001; Yin & Taylor, 2011).

northern Tibet, albeit with different magnitudes. The contraction direction was dominantly NNE-SSW to NE-SW. Both the models failed to produce E-W extensional strain in central Tibet (Figures 3b and 4b). In contrast, the Type 3 model at the late stage (i.e., after 70% of total convergence) gave rise to strong ESE-WNW extensional strains in central Tibet (Figure 5b-iv), which is a characteristic feature of the present-day deformation in Tibet (Figure 8a). Interestingly, this last stage of convergence had contraction deformations in north-eastern Tibet and the region between EHS and Sichuan basin. However, the E-W contraction in the region between EHS and Sichuan basin showed a transition to E-W extension down to the southern region of Burmese platelet. We have compared the strain-rate field of the Type 3 model experiment with the present-day geodetic strain rates (Figures 8a and 8b). Their remarkable match suggests the Type-3 model experiment provides the best approximation to the kinematic history of the Tibetan plateau.

#### 4.3. Orogen-Parallel Shear Deformations

We have developed a shear strain-rate ( $\dot{\epsilon}_{xy}$ ) map from the Type 3 model velocity field for the present-day Tibetan plateau. The map shows shear partitioning in laterally persistent orogen-parallel belts across model Tibet. Strong sinistral shear motion ( $-1.5 \times 10^{-8} \text{ yr}^{-1}$ ) localizes in northern Tibet, whereas dextral shear motion of nearly equal magnitude ( $1.5 \times 10^{-8} \text{ yr}^{-1}$ ) concentrates in southern Tibet (Figure 8c). The transition between sinistral and dextral shear occurs in central Tibet, where the magnitude of shear rate becomes low (Figure 8c).



The comparative analysis of the model results obtained from the three types of experiments suggest that the Type 3 model convincingly accounts for the effects of both lateral crustal heterogeneity and decreasing India-Asia indentation velocity in setting the present-day eastward crustal flow and the heterogeneous strain rate patterns in Tibet. We thus discuss the implications of our experimental study in the light of Type 3 model results in the following discussions.

#### 5. Discussion

#### 5.1. Tectonic Evolution of the Tibetan Plateau: Model Versus Nature

Our laboratory model results suggest that the high-velocity crustal-flows localized initially close to the leading edge of the Indian indenter (Figure 5a), and resulted in rapid topographic uplift in southern and central Tibet (Figures 5c i-ii) during the initial stage of fast India-Asia collision (5.5–4.5 cm/yr). In the course of collision the high-velocity front propagated north to north-eastward (Figures 5a and 5b). This kinematic change led to northward migration of the topographic uplift in Tibet. However, in the late stage of collision with slow rates (3.5 cm/yr) north-eastern and eastern Tibet became active to develop elevated topography. On the other hand, the geological records used to establish the topographic evolution of the Tibetan plateau are not straightforward, but pose several debated issues. A line of evidences has been put forward to show that the central and some parts of southern Tibet already attained their elevation of 3-4 km by 40 Ma to form plateau topography, referred to as proto-Tibetan plateau (Kapp et al., 2005; Murphy et al., 1997; Y. Li, Wang, et al., 2015; Wang et al., 2014). Several workers have proposed the collision between Lhasa and Qiangtang block during the Late Cretaceous-Paleogene time (Jolivet, 2015; Kapp et al., 2005; Murphy et al., 1997), prior to the India-Asia collision at 50 Ma as the responsible factor for the growth of this pre-existing plateau. In counter to this view, (Spicer et al., 2021), based on paleontological proxies for paleoaltimetry, have recently shown central Tibet as a low-elevation region between the Gangdese Mountain in the south and the Qiangtang terrain in the north, which was originally below the sea level during the Cretaceous-Paleogene time. From magnetostratigraphic and radiochronologic dating Fang et al. (2020) have provided a similar interpretation for central Tibet topography with a low elevation (<2.3 km) at 39.5 Ma, which gained a height of 3.5-4.5 km at around 26 Ma. On the other hand, Y. Li, Wang, et al. (2015) compiled the available thermochronological data to demonstrate that the central Tibetan plateau, in overall, continued to grow till  $\sim 20$  Ma. Their analysis also stressed upon the fact that the Tibetan plateau progressively shifted its uplift location from the central to the northern region. This northward migration of the plateau upliftment is consistent with our model observations. The spatio-temporal variations of model crustal flows gave rise to heterogeneous strain distributions in the entire history of the Tibetan plateau (Figure 5b). In the stage (~50 to  $19 \pm 3$  Ma) of high convergence rates (5.5–4.5 cm/vr) the NNE-SSW contraction determined the tectonic deformation patterns in the plateau (Figures 5b i-iii). When the convergence rate was reduced to 3.5 cm/ yr (the present-day average value) at  $19 \pm 3$  Ma the contractional mode of tectonic deformation started replacing by a distinct ESE-WNW extensional tectonics in southern and central Tibet (Figures 5b-iv and 8b) due to a topography-controlled gravity current (Copley & Mckenzie, 2007; Liang et al., 2013). This tectonic transition predicted from our scaled laboratory experiments is consistent with the deformation history of the Tibetan plateau. Geological imprints suggest that the E-W striking thrust faults in central Tibet were mostly active in the period of ~50 to ~22 Ma, and they subsequently became dormant. Active thrust faulting then shifted to localize preferentially in the southern and north-eastern margins (Y. Li, Wang, et al., 2015; Wang et al., 2014). After  $19 \pm 3$  Ma the onset of ESE-WNW extensional tectonics manifested in the form of N-S striking, low-angle normal faulting in central and southern Tibet (Ge et al., 2015; Kali et al., 2010; Lee et al., 2011; Y. Li, Wang, et al., 2015; McCallister et al., 2014; Styron et al., 2013). Most of the available field geological data suggest that this transition took place across the orogen over a timespan, broadly in middle Miocene. For example, the N-S stretching extensional grabens and normal faults were initiated in southern Tibet at 14-13 Ma (McCallister et al., 2014), ~13-12 Ma (Kali et al., 2010; Lee et al., 2011), ~11 Ma (Garzione et al., 2003), <12.5 Ma (Edwards & Harrison, 1997) etc. In further north, similar ages are reported, 16-12 Ma (Styron et al., 2013), 11-8 Ma (Woodruff et al., 2013), ~8 Ma (Harrison et al., 1995) and in northern Tibet it is ~13.5 Ma (Blisniuk et al., 2001). Recently, Bian, Gong, Zuza, et al. (2020) have suggested monotonically decreasing ages of rift developments across the orogen (from ca. 23 Ma at Leo Pargil to 2-3 Ma in the east at Cona). This observation is consistent with our present tectonic model, which produced GPE gradient-driven eastward lithospheric flows with reducing convergence rates. However, some of the recent studies have reported present-day thrust activities from Gangdese (Taylor et al., 2021) and Qimen Tagh areas (Wu et al., 2020, 2021; Yakovlev, 2015) of southern and central Tibet. These observations suggest that the thrust systems in central Tibet, although they are not major contributors to the ongoing Tibetan tectonics, are not totally inactive.

Our experimental model in the late stage, that is, from  $19 \pm 3$  Ma to the present day, produced a strong E-W crustal flows (~1.5 cm/yr) in central Tibet (Figure 8b). The model estimate suggests an extensional strain of 0.8 to  $2.2 \times 10^{-8}$  yr<sup>-1</sup> in this region (Figure 8b-ii), which fairly agrees with the geodetic estimates (1.6 to  $2.2 \times 10^{-8}$  yr<sup>-1</sup>, Ge et al., 2015). The time series analyses of our model crustal-flow and the strain pattern support a two-stage evolution of the Tibetan plateau: (a) a prolonged period of contraction since the beginning of collision to as late as  $19 \pm 3$  Ma, which was coupled with (b) a phase of wide-spread horizontal extension in the mid-Miocene time ( $19 \pm 3$  Ma) preferentially in central Tibet (Maiti et al., 2020; Maiti & Mandal, 2021; Wang et al., 2014).

#### 5.2. Contractional Deformations in the NE Flank of Tibet

The model experiments suggest that north-eastern and eastern Tibet in adjacent to the Sichuan basins localized southward crustal flows at high rates (~3.5 cm/yr) in the late stage (i.e., after 70% convergence) of India-Asia collision after ~22 Ma (Figures 5a iii-iv). Model topographic analysis reveals that these regions gained significant topographic elevations after  $19 \pm 3$  Ma (Figure 5c-iii). The strain-rate plot also indicates NE-SW contractional deformations in north-eastern Tibet and an E-W contraction near Sichuan basins during this late stage (Figures 5b iii-iv). Using thermochronological data several workers have constrained the initiation of upliftment/exhumation of the Longmen Shan thrust (LMST, Figure 7a, boundary between Sichuan basin and eastern Tibet) at around 22 Ma (Wilson & Fowler, 2011; Xu & Kamp, 2000). Although some studies indicated the existence of early elevated topography in Longmen Shan (X. Chen et al., 2003; Wallis et al., 2003), it has been suggested from local structural analyses and river incision data that the rapid exhumation and high topographic elevation occurred in this region dominantly during 12-10 Ma (Clark et al., 2005; E. Wang et al., 2012; Xu & Kamp, 2000). Moreover, some thrusts (i.e., Ganzi-Litang thrust belt, GLT) in the western part of the Longmen Shan region showed rapid exhumation during 20-16 Ma (Figure 7a). The deformation history of these thrust belts show an eastward migration of their activities from GLT in the west to Longmen Shan in the east. All these data reveal a Miocene uplift event in the eastern-most Tibetan region, which strongly supports our model results. On the other hand, some studies suggested the Qaidam basin in north-eastern Tibet to have developed its elevated topography during 40 to 22 Ma (Song et al., 2020; E. Wang et al., 2012; Yin et al., 2008). However, the existence of such elevated early topography is still debated (Kent-Corson et al., 2009; L. Wang, Cheng, et al., 2021). On further north-east, in the Qilian Shan region several NE-directed thrust faults (i.e., Nanshan Thrust belt, NST; Eastern Qlianshan-Nanshan thrust belt, EONT; Northern Oilian Shan thrust belt, NOLT) became active in between 33 and 9 Ma (Yin et al., 2002). Among them, NQLT is still active at present day (S. Yang et al., 2007; Figure 7a). These observations suggest that the major uplift in the NE region occurred in between 24 and 9 Ma, although some parts of Oilian Shan region started to gain their elevation since 40 Ma (An et al., 2020; Bian, Gong, Chen, et al., 2020; Clark et al., 2010; B. Li et al., 2020; L. Wang, Cheng, et al., 2021). Our model also indicates that a small amount of the contractional deformation propagated northward to affect north-eastern Tibet during the fast-collision period (Figures 5a-ii, 5b-ii), but the major contractional deformations in the extreme north-eastern margin of Tibet was a late  $(19 \pm 3 \text{ Ma to present-day})$  tectonic event (Figures 5b iii-iv). Several studies, however, have shown a complex pattern of topographic uplift in Tibet due to the presence of numerous pre-collision arcs and mechanically weaker lithospheric segments, as well as reactivation of suture zones (Zuza & Yin, 2017). For example, numerical modeling of Bian, Gong, Chen, et al. (2020) suggests that Qilian Shan region underwent early Cenozoic uplift preferentially as a consequence of its weaker lithospheric rheology. L. Wang, Cheng, et al. (2021) have recently estimated the early Cenozoic topographic elevation of Qilian Shan varying between 0.4 and 1.5 km, which is relatively higher than that in Eastern Kunlun Shan (0.4-1 km) located to its south. However, during Neogene time Eastern Kunlun Shan had tectonic uplift more than Qilian Shan (located in far north-eastern Tibet). With the present-day topographic elevation of  $\sim$  3–4 km in Eastern Kunlun Shan and  $\sim$  2–3 km in Qilian Shan, it is clear that, despite the occurrence of some pre-existing topographic relief in north-eastern Tibet the topographic upliftment

progressed from south to north, where GPE controlled north-east wards crustal flows might have played an important role, as inferred by L. Wang, Cheng, et al. (2021).

The present-day NE-SW contractions in north-eastern Tibet occurs at a strain-rate of 2 to  $3.2 \times 10^{-8}$  yr<sup>-1</sup>, estimated from GPS data (Gan et al., 2007). Our model estimate yields a similar order of strain-rates  $(2 \times 10^{-8} \text{ yr}^{-1})$ . The geodetic observation also indicates that the contraction rates in the region between EHS and Sichuan basin are measured as 3 to  $4 \times 10^{-8} \text{ yr}^{-1}$  (Gan et al., 2007), which fairly matches with our model estimates, 3.2 to  $4.2 \times 10^{-8} \text{ yr}^{-1}$  (Figures 8a and 8b). Based on the correlation between the model and natural data, we suggest that the contractional deformation in NE Tibet is driven by the NE-directed crustal flows from the plateau interior. On the other hand, in eastern Tibet the eastward crustal-flow encountered a strong resistance from the Sichuan basin to localize a contractional strain field. Clark and Royden (2000) proposed a mechanical model to explain the development of topography in eastern Tibet. According to their model, the topographic pressure driven lower crustal flows determined the topographic elevation. Our model also suggests GPE controlled flows, but on a lithospheric scale, as the underlying factors for topographic growth in eastern Tibet.

#### 5.3. The Present-Day Crustal Velocity Field and the Plateau Topography

Type-3 model estimates yield the present-day velocity magnitudes and their directions in southern ( $\sim$ 3.5 cm/yr), central ( $\sim$ 3.2 cm/yr), and north-eastern Tibet (0.5–1 cm/yr) (Figures 5a-iv, 6c), which match well with the GPS velocities,  $\sim$ 3.8 cm/yr,  $\sim$ 3.5 cm/yr, and 0.5–1.5 cm/yr in the respective regions (Gan et al., 2007; Liang et al., 2013; Zhang et al., 2004). In addition, the model reproduces a spectacular clockwise rotation of the velocity vectors (1–2 cm/yr) around EHS, as predicted from GPS studies (Zhang et al., 2004). The present-day topographic gradients in the north-south (AB), east-west (CD) sections as well as in north-eastern and south-eastern model Tibet also show a good match with those obtained from the digital elevation maps of Tibet (Figure 7b). The finding validates our model interpretation that the eastward first-order topographic gradient drives the present-day west to east crustal flows, as shown in earlier models (Cook & Royden, 2008; Copley & Mckenzie, 2007; Klemperer, 2006; Liu & Yang, 2003).

#### 5.4. Eastward "Glacier-Like" Flow and Its Implications

We calculated the E-W component of crustal-flow velocities from the Type 3 model that represents the present-day tectonics of Tibet. Its N-S variation describes a parabolic velocity profile with the maximum magnitude located in the central part and the minimum in the northern and southern margins of Tibet (Figure 9a). The parabolic velocity profile typically represents a Poiseuille type flow, implying that the eastward flow is driven by a pressure gradient (Yin & Taylor, 2011). Yin and Taylor (2011) showed that, in the last 15 Ma the continental lithosphere (on a horizontal scale of 1,000 km) in central Tibet flowed eastward like a fluid mass, between two rigid walls, relatively moving towards each other (Bajolet et al., 2015; Fournier et al., 2004). They compared their model observations with existing GPS data, and proposed horizontal shear applied at the base of the Tibetan mantle lithosphere or lateral spreading of the thickened Tibetan lithosphere as a driving factor for the late Cenozoic deformation in central Tibet. Our model suggests that the first-order E-W topographic gradient induced a pressure difference, which eventually acted as the driving force for this flow, as proposed by Copley and Mckenzie (2007) for the gravitational spreading of crustal materials in southern and south-eastern Tibet. The eastward-flow began only after the indentation velocity decreased to the present-day average of 3.5 cm/yr (Figures 5a-iii-iv). A comparison of the model E-W velocity fields with the GPS estimates of Zhang et al. (2004), also suggest that their parabolic profile is a characteristic representation of the present-day kinematics in the central Tibetan plateau (Figure 9b). The model experiments reveal another interesting fact that the eastward flows strengthen their velocity magnitudes in the east direction, 0.5 cm/yr to nearly 2 cm/yr, which conforms to GPS observations (Figure 9c). Both the model and GPS data show the north-directed crustal flows with decreasing magnitudes from south to north, 3.5 cm/yr to  $\sim 1 \text{ cm/yr}$  (Figure 9d).

Our model results discussed above suggest that the decelerating collision kinematics set a Poiseuille-type eastward crustal-flow in central Tibet, where the Tarim basin in the north and the Himalaya in the south acted as the guiding walls (Figures 9a and 9b). This horizontal flow resulted in sinistral shear strain in the





**Figure 9.** (a) East-west component of the total velocity obtained from Type 3 model at 100% convergence ( $\epsilon$ ), equivalent to present-day Tibet. Note that the velocity profiles describe a parabolic curve, suggesting a Poiseuille type flow in the Tibetan plateau driven by a pressure gradient. (b) Comparison of the eastward component of model velocity (solid circle) with the GPS data (solid square) from Zhang et al. (2004) along the green, red and blue lines (marked in Figure 1a). (c) Comparison of the trend of increasing eastward velocity between experiment and GPS data along CD (Figure 1c). (d) Decreasing northward velocity in present-day Tibet, obtained from experiment and GPS data along the green, red and blue lines (marked in Figure 1a).

north and dextral shear strain in southern Tibet (Figure 8c). The two shear fields of opposing shear senses explain most of the strike-slip faults (Taylor & Yin, 2009) and their motion in southern and northern Tibet (Figure 8c). Yin and Taylor (2011) combined field observations with their mechanical models (localized basal shear or lateral gravity spreading) to demonstrate the kinematics of these two shear domains, and explained the occurrence of right and left strike-slip faults in central Tibet as a conjugate strike-slip fault system (Figure 8c). Their model, however, does not account for the origin of non-conjugate right- and left-slip faults on either side of the conjugate fault system. Our model Tibet, on the other hand, produces two shear fields in the extreme southern and northern parts of Tibet, with a transition zone of little or no shear in the central part. We suggest that the conjugate faults with opposing slip senses have formed in this transition region dominated by symmetrical eastward flows, whereas non-conjugate left- and right-slip strike faults in the shear dominated fields of its northern and southern sides, respectively.

#### 5.5. Role of Lateral Crustal Heterogeneities in Tibetan Tectonics

In the laboratory model experiments the rigid Tarim basin lying on north of Tibet acted as a mechanical barrier to the northward crustal flows in the Tibetan plateau, and redirected the flows in the NE direction (Figures 10a and 10b). Moreover, this resistance forced the Tibetan crust to uplift at higher rates in its western





**Figure 10.** A cartoon presentation of the three-stage evolution of crustal-flow pattern in the Tibetan plateau with decreasing Indian indentation velocity  $(I_{\text{DEC}})$ . (a) During high  $I_{\text{DEC}} = 5.5$  cm/yr, the collision produced an overall NNE- directed crustal flow in Tibet. At this stage N-S shortening and the topographic uplift localized mostly in the southern region. Note that, bold red arrow denotes overall crustal flow pattern and brown bold arrows (double head) indicate contractional deformations. (b) With continuous northward indentation  $(I_{\text{DEC}} = 4.5 \text{ cm/yr})$  crustal shortening reached northern Tibet, where the flows encountered resistance from rigid Tarim basin, and were deflected towards northeast. At this stage the western and central Tibet uplifted at relatively higher rates, compared to eastern Tibet, creating an eastward topographic gradient (0.1°). (c) At  $19 \pm 3$  Ma ( $\epsilon = 70\%$ ), decrease of  $I_{\text{DEC}}$  to the present average (3.5 cm/yr), resulted in gravitational collapse of the plateau. The eastward topographic gradient directed the crustal flow towards east to initiate the E-W extensional tectonics, as observed in the present-day configuration. Note that, during this stage the eastward crustal flow followed a Poiseuille type velocity profile, giving rise to sinistral and dextral shear in northern and southern Tibet, respectively.

side, relative to its eastern counterpart (Figure 7c). Such a differential uplift eventually developed first-order topography with an eastward surface slope. Interestingly, Tibet still maintains an eastward slope of  $0.03^{\circ}$  (Figure 7b). Numerical models of Yang and Liu (2013) also produced results similar to that observed in our viscous analogue model, where Tarim basin in the north produced higher topographic upliftment in the western Tibet, compared to east. Their model also predicted that the zone of crustal uplift propagated from south to north in the early stages and west to east in the later stages. However, their model Tibetan plateau attained a maximum topographic elevation of 5 km in the last 5-0 Ma of India-Asia collision. In contrast, the same maximum elevation was attained in our model by  $19 \pm 3$  Ma, as also estimated from various field observations (DeCelles et al., 2007; Fang et al., 2020).

Our model experiment strongly supports the hypothesis proposed by several earlier workers that the eastward flows encountered a resistance from the mechanically strong Sichuan basin in eastern Tibet (Clark et al., 2005; Cook & Royden, 2008; Penney & Copley, 2021). The Sichuan basin thus acted as a crucial factor in forcing the crustal flows to take a clockwise turn around the EHS (Figure 10c). Interestingly, this experimental finding holds an excellent agreement with previous numerical India-Asia collision models (Cook & Royden, 2008; Pusok & Kaus, 2015; Yang & Liu, 2013). In addition, the present model allows us to propose that the fan-like flow divergence in the Burmese plate region occurred dominantly in a later stage (after ~19 Ma) of the India-Asia collision when the convergence rate reduced to 3.5 cm/yr. The interaction of Tarim basin with the Tibetan plateau in the course of India-Asia collision has been quite complex, showing both underthrusting of Tarim and strike-slip faulting (S. Wang, Chen, et al., 2021; Wei et al., 2013). From teleseismic studies Kao et al. (2001) investigated lateral structural changes around Tarim and inferred that the basin behaved to be rigid with little shortening. They also found the Moho discontinuity of Tarim dips gently toward the south to ~50 km under the Kunlun foreland, and proposed for a complex pattern of lithospheric collision than the simple configuration of one block subducting beneath another. By correlating seismic reflectors with the corresponding dated outcropping sections, S. Wang, Chen, et al. (2021) have shown that rigid Tarim lithosphere is underthrusting beneath the western Kunlun Shan region of Tibet since ~30 Ma at an average rate of ~8.18 mm/yr. This rate comprises shortening rate of ~3.8–5.7 mm/yr and latitudinal propagation (northward) rate of the western Kunlun Shan ~4.4–2.5 mm/ yr. In our experiments the model Tibetan crustal masses had a tendency to flow over the Tarim block at the late stage (after  $19 \pm 3$  Ma), suggesting the Tarim block relatively underthrusting against the Tibet (Figure S4-b in Supporting Information S1).

## 6. Limitations and Future Scopes

Our laboratory modeling considers a two-layer mechanical structure of the Himalaya-Tibet system, consisting of a top high-viscosity layer (crustal lithosphere) upon a relatively low-viscosity layer (mantle region). The model crustal lithosphere is treated as a single viscous sheet, scaled to represent the depth-averaged density and viscosity of a 40 km thick crust. The model thus reproduces the overall continental-scale deformation in Tibet and its first-order topography produced during collision, as in the case of earlier TVS approximations (England & McKenzie, 1982). This model excludes higher-order discontinuities (i.e., faults, folds etc.) and their topographic effects, which is a limitation of the present study. In addition, we have entirely constrained the model boundary conditions kinematically to incorporate time dependent variations in India-Asia convergence velocities based on the available geological records. However, changes in plate motion are necessarily driven by changes in one or more driving or resisting forces. Iaffaldano et al. (2006) showed that, for large mountain belts and plateaus the topographic loads consume a significant amount of the driving tectonic forces by increasing frictional resistance between the overriding and subducting plates. The changing force-balance condition can thereby govern the long-term evolution of plate motion. Our present model study, however, does not account for the dynamics responsible for reducing convergence rate in the course of the India-Asia collision, which is also a limitation.

Some workers suggested that rigid Tarim block at the northern margin and the Sichuan basin in the eastern flank are pushed northward and eastward, respectively in the course of the India-Asia collision (Pan & Shen, 2017; S. Wang, Chen, et al., 2021). In fact, the GPS observations support this hypothesis (Liang et al., 2013; Zhang et al., 2004). However, there is a considerable uncertainty with the exact timing of such movement. Moreover, Meyer et al. (1998) showed that the northern boundary of Tibet adjacent to the Altyn Tagh Fault region remained stable since the beginning of the India-Asia collision. Clark et al. (2010) have also shown that the northern and north-eastern margins of Tibet remained nearly at the same position relative to stable Eurasia in the entire period of India-Asia collision, barring a little northward movement. Considering this kinematics of the northern margin on geological time scales, we have chosen to impose a fixed-wall condition for these rigid blocks. There is a scope of the present experimental study to explore the effects of their present-day movement, as predicted from GPS velocities, on the crustal flow pattern in Tibet.

Our experimental models reproduce the horizontal strain rate tensors grossly in agreement with the geodetic strain rates in Tibet. However, several studies have shown contraction deformations at the southern margin of Tibet as a manifestation of transient elastic strain accumulation around the Himalayan thrust faults, that is, thrusting on the range margin (e.g., Stevens & Avouac, 2015). These transient elastic effects are not accounted for in our experiments, which is evidently a limitation in our study. The present model thus does not fully capture the governing dynamics of Tibet and some deformational observations, such as dominantly ESE-WNW permanent extensional strains with little or no NE-SW contraction in southern Tibet, as revealed from earthquake focal mechanism solutions (Armijo et al., 1986; Copley et al., 2011; Elliott et al., 2010; Styron et al., 2015). Also, our model Tibet considers a flat initial topography. However, there is an ample evidence for the existence of pre-collisional topographic elevation in central (i.e., Lhasa and Qiantang blocks) and north-eastern Tibet (Qilian Shan), as discussed earlier. These high-topographic zones are expected to undergo uplift at lower rates due to their greater GPE during the collision. Consequently, crustal shortening would preferentially localize more intensely in the low-lying areas. There is a scope for future studies to validate this hypothesis experimentally by incorporating pre-existing topography and crustal heterogeneities in the initial model setup. Despite all these limitations, our Tibet model successfully demonstrates the development of the first-order E-W Tibetan topographic slopes, and their subsequent collapse with decreasing Indian convergence rates, setting in eastward crustal flow in the late phase of India-Asia collision (Figures 7b and 10). An agreement of the model predicted present-day crustal-flow velocities (Figure 6c) and the overall strain tensor fields (Figures 8a and 8b) with actual natural data provide additional strengths to this study.

In a recent study (Webb et al., 2017) it has been shown that the change in slab dynamics, anchoring of the Indian continental slab during 30 to 25 Ma and subsequent slab breakoff in the last 13 Ma, has critically influenced the Himalayan tectonics. The slab breakoff event resulted in a shift of crustal shortening from deep inside the orogenic wedge in the early phase to a rapid rise of the Mountain in the last 13 Ma years. In the present model we have considered a simple steady underthrust geometry, without any slab steepening, break-off, or flattening in the experimental setup. Recent numerical models of Dasgupta et al. (2021) allow us to stipulate that unsteady slab-dip (i.e., steepening or flattening) would be an additional factor in determining the contractional versus extensional tectonics of the overriding plate. Thus, this issue opens up a further scope for advancing the present Tibet model with an unsteady slab underthrusting.

## 7. Conclusions

The laboratory model experiments lead us to the following conclusions. (a) Steady-state kinematics of India-Asia collision at either high or low convergence rates would not produce the present-day crustal flow and deformation patterns in the Tibetan plateau, as predicted from GPS observations. (b) We interpret the onset of eastward crustal flow and related ESE-WNW extension in Tibet as a consequence of decelerating Indian indentation velocity ( $I_{\text{DEC}}$ ), from ~5.5 cm/year at ~50 Ma to ~3.5 cm/year at 19 ± 3 Ma in the India-Asia collision history. (c) The rigid Tarim block, located in the northern boundary of Tibet has played a critical role (Figure 10) in developing an eastward first-order topographic gradient by  $19 \pm 3$  Ma (Figure 7c). This topographic gradient later dictated the crustal material to flow in the east direction, resulting in a transition of NNE-SSW contraction to ESE-WNW extensional tectonics and formation of widespread nearly N-S trending grabens in southern and central Tibet when  $I_{\rm DEC}$  decreased to ~3.5 cm/yr at 19 ± 3 Ma. (d) The present model also suggests that the eastward crustal flow eventually encountered the rigid Sichuan basin in eastern Tibet (Figure 10c), which forced the flow to take a turn (clockwise rotation) in the southern direction. (e) The orogen-parallel velocity component in our model Tibet follows approximately a parabolic variation from its southern to northern boundary, which broadly agrees with the GPS velocity profile. We thus propose Poiseuille-type kinematics for the orogen-parallel flow component, driven by a pressure difference due to the west to east topographic gradient. This Poiseuille-type flow explains the localization of sinistral and dextral strike-slip faults preferentially in the northern and the southern parts of Tibet. Finally, our laboratory model estimates of extensional and contractional strain rates: 0.8 to  $2.2 \times 10^{-8}$  yr<sup>-1</sup> and 3.2 to  $4.2 \times 10^{-8}$  yr<sup>-1</sup> confirm the GPS-based strain rates in central Tibet and its eastern flank, respectively.

## **Data Availability Statement**

The authors confirm that all the data supporting the findings of this study are available within the article. GPS and strain-rate data used for comparison of model results are available in Zhang et al. (2004), Gan et al. (2007), and Zheng et al. (2017).

## References

An, K., Lin, X., Wu, L., Yang, R., Chen, H., Cheng, X., et al. (2020). An immediate response to the Indian-Eurasian collision along the northeastern Tibetan Plateau: Evidence from apatite fission track analysis in the Kuantan Shan-Hei Shan. *Tectonophysics*, 774(2019), 228278. https://doi.org/10.1016/j.tecto.2019.228278

Armijo, R., Tapponnier, P., Mercier, J. L., & Han, T.-L. (1986). Quaternary extension in southern Tibet: Field observations and tectonic implications. *Journal of Geophysical Research*, 91, 13803–13872. https://doi.org/10.1029/jb091ib14p13803

#### Acknowledgments

We greatly benefitted from the insightful comments and constructive suggestions by Andrew Zuza, an anonymous reviewer and Associate Editor Laurent Husson in revising our study. We also thank Editor Claudio Faccenna for providing us a guideline with several important inputs to improve the manuscript. This work has been supported by the DST-SERB through J. C. Bose fellowship (SR/S2/JCB-36/2012) to N. Mandal. G. Maiti and A. Roy gratefully acknowledge CSIR, India for awarding them senior and junior research fellowships, respectively. J. Sen is supported with a junior research fellowship from DST-INSPIRE, India



- Bajolet, F., Chardon, D., Martinod, J., Gapais, D., & Kermarrec, J. J. (2015). Synconvergence flow inside and at the margin of orogenic plateaus: Lithospheric-scale experimental approach. *Journal of Geophysical Research: Solid Earth*, *120*, 6634–6657. https://doi. org/10.1002/2015JB012110
- Bajolet, F., Replumaz, A., & Lainé, R. (2013). Orocline and syntaxes formation during subduction and collision. *Tectonics*, 32, 1529–1546. https://doi.org/10.1002/tect.20087
- Beaumont, C., Jamieson, R. A., Nguyen, M. H., & Lee, B. (2001). Himalayan tectonics explained by extrusion of a low-viscosity crustal channel coupled to focused surface denudation. *Nature*, 414, 738–742. https://doi.org/10.1038/414738a
- Bian, S., Gong, J., Chen, L., Zuza, A. V., Chen, H., Lin, X., et al. (2020). Diachronous uplift in intra-continental orogeny: 2D thermo-mechanical modeling of the India-Asia collision. *Tectonophysics*, 775(2019), 228310. https://doi.org/10.1016/j.tecto.2019.228310
- Bian, S., Gong, J., Zuza, A. V., Yang, R., Tian, Y., Ji, J., et al. (2020). Late Pliocene onset of the Cona rift, eastern Himalaya, confirms eastward propagation of extension in Himalayan-Tibetan orogen. *Earth and Planetary Science Letters*, 544, 116383. https://doi.org/10.1016/j. epsl.2020.116383
- Bird, P. (1991). Lateral extrusion of lower crust from under high topography, in the isostatic limit. *Journal of Geophysical Research*, 96(B6), 275–286. https://doi.org/10.1029/91jb00370
- Bischoff, S. H., & Flesch, L. M. (2018). Normal faulting and viscous buckling in the Tibetan Plateau induced by a weak lower crust. *Nature Communications*, 9. https://doi.org/10.1038/s41467-018-07312-9
- Blisniuk, P. M., Hacker, B. R., Glodny, J., Ratschbacher, L., Bi, S., Wu, Z., et al. (2001). Normal faulting in central Tibet since at least 13.5 Myr ago. Nature, 412, 628–632. https://doi.org/10.1038/35088045
- Chen, L., Capitanio, F. A., Liu, L., & Gerya, T. V. (2017). Crustal rheology controls on the Tibetan plateau formation during India-Asia convergence. *Nature Communications*, 8. https://doi.org/10.1038/ncomms15992
- Chen, X., Yin, A., Gehrels, G. E., Cowgill, E. S., Grove, M., Harrison, T. M., & Wang, X. F. (2003). Two phases of Mesozoic north-south extension in the eastern Altyn Tagh range, northern Tibetan Plateau. *Tectonics*, 22(5), 1053. https://doi.org/10.1029/2001TC001336
- Chung, S. L., Lo, C. H., Lee, T. Y., Zhang, Y., Xie, Y., Li, X., et al. (1998). Diachronous uplift of the Tibetan plateau starting 40 Myr ago. *Nature*, 394, 769–773. https://doi.org/10.1038/29511
- Clark, M. K. (2012). Continental collision slowing due to viscous mantle lithosphere rather than topography. *Nature*, 483(7387), 74–77. https://doi.org/10.1038/nature10848
- Clark, M. K., Farley, K. A., Zheng, D., Wang, Z., & Duvall, A. R. (2010). Early Cenozoic faulting of the northern Tibetan Plateau margin from apatite (U-Th)/He ages. *Earth and Planetary Science Letters*, 296(1–2), 78–88. https://doi.org/10.1016/j.epsl.2010.04.051
- Clark, M. K., House, M. A., Royden, L. H., Whipple, K. X., Burchfiel, B. C., Zhang, X., & Tang, W. (2005). Late Cenozoic uplift of southeastern Tibet. *Geology*, 33, 525–528. https://doi.org/10.1130/G21265.1
- Clark, M. K., & Royden, L. H. (2000). Topographic ooze: Building the eastern margin of Tibet by lower crustal flow. *Geology*, 28, 703–706. https://doi.org/10.1130/0091-7613(2000)028<0703:tobtem>2.3.co;2
- Cook, K. L., & Royden, L. H. (2008). The role of crustal strength variations in shaping orogenic plateaus, with application to Tibet. Journal of Geophysical Research, 113, B08407. https://doi.org/10.1029/2007JB005457
- Cooper, F. J., Hodges, K. V., Parrish, R. R., Roberts, N. M. W., & Horstwood, M. S. A. (2015). Synchronous N-S and E-W extension at the Tibet-to-Himalaya transition in NW Bhutan. *Tectonics*, 34, 1375–1395. https://doi.org/10.1002/2014TC003712
- Copley, A. (2012). The formation of mountain range curvature by gravitational spreading. *Earth and Planetary Science Letters*, 351–352, 208–214. https://doi.org/10.1016/j.epsl.2012.07.036
- Copley, A., Avouac, J. P., & Royer, J. Y. (2010). India-Asia collision and the Cenozoic slowdown of the Indian plate: Implications for the forces driving plate motions. *Journal of Geophysical Research*, 115, B03410. https://doi.org/10.1029/2009JB006634
- Copley, A., Avouac, J. P., & Wernicke, B. P. (2011). Evidence for mechanical coupling and strong Indian lower crust beneath southern Tibet. *Nature*, 472, 79–81. https://doi.org/10.1038/nature09926
- Copley, A., & Mckenzie, D. (2007). Models of crustal flow in the India-Asia collision zone. *Geophysical Journal International*, 169, 683–698. https://doi.org/10.1111/j.1365-246X.2007.03343.x
- Craig, T. J., Kelemen, P. B., Hacker, B. R., & Copley, A. (2020). Reconciling geophysical and petrological estimates of the thermal structure of southern Tibet. *Geochemistry, Geophysics, Geosystems*, 21(8), e2019GC008837. https://doi.org/10.1029/2019GC008837
- Dasgupta, R., Mandal, N., & Lee, C. (2021). Controls of subducting slab dip and age on the extensional versus compressional deformation in the overriding plate. *Tectonophysics*, 801(2020), 228716. https://doi.org/10.1016/j.tecto.2020.228716
- DeCelles, P. G., Quade, J., Kapp, P., Fan, M., Dettman, D. L., & Ding, L. (2007). High and dry in central Tibet during the Late Oligocene. Earth and Planetary Science Letters, 253, 389–401. https://doi.org/10.1016/j.epsl.2006.11.001
- DeCelles, P. G., Robinson, D. M., & Zandt, G. (2002). Implications of shortening in the Himalayan fold-thrust belt for uplift of the Tibetan Plateau. *Tectonics*, 21. 12-1–12-25. https://doi.org/10.1029/2001TC001322
- Dewey, J. F., Shackleton, R. M., Chengfa, C., & Yiyin, S. (1988). The tectonic evolution of the Tibetan Plateau. Philosophical Transactions of the Royal Society of London. Series A, Mathematical and Physical Sciences, 327, 379–413. https://doi.org/10.1098/rsta.1988.0135
- Edwards, M. A., & Harrison, T. M. (1997). When did the roof collapse? Late Miocene north-south extension in the high Himalaya revealed by Th-Pb monazite dating of the Khula Kangri granite. *Geology*, 25, 543–546. https://doi.org/10.1130/0091-7613(1997)025<054 3:WDTRCL>2.3.CO;2
- Elliott, J. R., Walters, R. J., England, P. C., Jackson, J. A., Li, Z., & Parsons, B. (2010). Extension on the Tibetan plateau: Recent normal faulting measured by InSAR and body wave seismology. *Geophysical Journal International*, 183(2), 503–535. https://doi. org/10.1111/j.1365-246X.2010.04754.x
- England, P., & Houseman, G. (1986). Finite strain calculations of continental deformation: 2. Comparison with the India-Asia Collision Zone. *Journal of Geophysical Research*, 91, 3664–3676. https://doi.org/10.1029/jb091ib03p03664
- England, P., & Houseman, G. (1989). Extension during continental convergence, with application to the Tibetan Plateau. Journal of Geophysical Research, 94, 17561. https://doi.org/10.1029/jb094ib12p17561
- England, P., & McKenzie, D. (1982). A thin viscous sheet model for continental deformation. *Geophysical Journal of the Royal Astronomical Society*, 70, 295–321. https://doi.org/10.1111/j.1365-246X.1982.tb04969.x
- Fang, X., Dupont-Nivet, G., Wang, C., Song, C., Meng, Q., Zhang, W., et al. (2020). Revised chronology of central Tibet uplift (Lunpola Basin). *Science Advances*, 6. https://doi.org/10.1126/sciadv.aba7298
- Flesch, L. M., Haines, A. J., & Holt, W. E. (2001). Dynamics of the India-Eurasia collision zone. Journal of Geophysical Research, 106, 16435–16460. https://doi.org/10.1029/2001jb000208
- Fournier, M., Jolivet, L., Davy, P., & Thomas, J. C. (2004). Backarc extension and collision: An experimental approach to the tectonics of Asia. *Geophysical Journal International*, 157, 871–889. https://doi.org/10.1111/j.1365-246X.2004.02223.x

- Galland, O., Bertelsen, H. S., Guldstrand, F., Girod, L., Johannessen, R. F., Bjugger, F., et al. (2016). Application of open-source photogrammetric software MicMac for monitoring surface deformation in laboratory models. *Journal of Geophysical Research: Solid Earth*, 121, 2852–2872. https://doi.org/10.1002/2015JB012564
- Gan, W., Zhang, P., Shen, Z. K., Niu, Z., Wang, M., Wan, Y., et al. (2007). Present-day crustal motion within the Tibetan Plateau inferred from GPS measurements. *Journal of Geophysical Research*, 112, B08416. https://doi.org/10.1029/2005JB004120
- Garzione, C. N., DeCelles, P. G., Hodkinson, D. G., Ojha, T. P., & Upreti, B. N. (2003). East-west extension and Miocene environmental change in the southern Tibetan plateau: Thakkhola graben, central Nepal. *Bulletin of the Geological Society of America*, *115*, 3–20. https://doi.org/10.1130/0016-7606(2003)115<0003:EWEAME>2.0.CO;2
- Ge, W. P., Molnar, P., Shen, Z. K., & Li, Q. (2015). Present-day crustal thinning in the southern and northern Tibetan Plateau revealed by GPS measurements. *Geophysical Research Letters*, 42, 5227–5235. https://doi.org/10.1002/2015GL064347
- Ghosh, D., Maiti, G., Mandal, N., & Baruah, A. (2020). Cold plumes initiated by Rayleigh-Taylor instabilities in subduction zones, and their characteristic volcanic distributions: The role of slab dip. *Journal of Geophysical Research: Solid Earth*, *125*, e2020JB019814. https://doi.org/10.1029/2020JB019814

Girod, L. (2012). The use of SfM technologies in geosciences: Micmac for geologists Ecole Nationale des Sciences Geographiques - ENSG.

- Harrison, T. M., Copeland, P., Kidd, W. S. F., & Lovera, O. M. (1995). Activation of the Nyainqentanghla Shear Zone: Implications for uplift of the southern Tibetan Plateau. *Tectonics*, 14, 658–676. https://doi.org/10.1029/95TC00608
- Houseman, G., & England, P. (1986). Finite strain calculations of continental deformation: 1. Method and general results for convergent zones. Journal of Geophysical Research, 91, 3651–3663. https://doi.org/10.1029/jb091ib03p03651
- Houseman, G., & England, P. (1993). Crustal thickening versus lateral expulsion in the Indian-Asian continental collision. Journal of Geophysical Research, 98, 12233–12249. https://doi.org/10.1029/93jb00443
- Huang, W., Van Hinsbergen, D. J. J., Lippert, P. C., Guo, Z., & Dupont-Nivet, G. (2015). Paleomagnetic tests of tectonic reconstructions of the India-Asia collision zone. *Geophysical Research Letters*, 42, 2642–2649. https://doi.org/10.1002/2015GL063749
- Hubbert, M. K. (1937). Theory of scale models as applied to the study of geologic structures. *Bulletin of the Geological Society of America*, 48(10), 1459–1520. https://doi.org/10.1130/GSAB-48-1459
- Huyghe, P., Galy, A., Mugnier, J. L., & France-Lanord, C. (2001). Propagation of the thrust system and erosion in the Lesser Himalaya: Geochemical and sedimentological evidence. *Geology*, 29. https://doi.org/10.1130/0091-7613(2001)029<1007:POTTSA>2.0.CO;2

Iaffaldano, G., Bunge, H. P., & Dixon, T. H. (2006). Feedback between mountain belt growth and plate convergence. *Geology*, *34*, 893–896. https://doi.org/10.1130/G22661.1

- Ingalls, M., Rowley, D. B., Currie, B., & Colman, A. S. (2016). Large-scale subduction of continental crust implied by India-Asia mass-balance calculation. Nature Geoscience, 9, 848–853. https://doi.org/10.1038/ngeo2806
- Jiménez-Munt, I., Fernàndez, M., Vergés, J., & Platt, J. P. (2008). Lithosphere structure underneath the Tibetan Plateau inferred from elevation, gravity and geoid anomalies. Earth and Planetary Science Letters, 267, 276–289. https://doi.org/10.1016/j.epsl.2007.11.045
- Johnson, M. R. W. (2002). Shortening budgets and the role of continental subduction during the India Asia collision. *Earth-Science Reviews*, 59, 101–123. https://doi.org/10.1016/S0012-8252(02)00071-5
- Jolivet, M. (2015). Mesozoic tectonic and topographic evolution of Central Asia and Tibet: A preliminary synthesis. In *Geological society special publication* (Vol. 427, pp. 19–55). https://doi.org/10.1144/SP427.2
- Kali, E., Leloup, P. H., Arnaud, N., Mahéo, G., Liu, D., Boutonnet, E., et al. (2010). Exhumation history of the deepest central Himalayan rocks, Ama Drime range: Key pressure-temperature-deformation-time constraints on orogenic models. *Tectonics*, 29, TC2014. https:// doi.org/10.1029/2009tc002551
- Kao, H., Gao, R., Rau, R. J., Shi, D., Chen, R. Y., Guan, Y., & Wu, F. T. (2001). Seismic image of the Tarim basin and its collision with Tibet. Geology, 29(7), 575–578. https://doi.org/10.1130/0091-7613(2001)029<0575:SIOTTB>2.0.CO;2
- Kapp, P., Yin, A., Harrison, T. M., & Ding, L. (2005). Cretaceous-Tertiary shortening, basin development, and volcanism in central Tibet. Bulletin of the Geological Society of America, 117(7–8), 865–878. https://doi.org/10.1130/B25595.1
- Kent-Corson, M. L., Ritts, B. D., Zhuang, G., Bovet, P. M., Graham, S. A., & Chamberlain, C. P. (2009). Stable isotopic constraints on the tectonic, topographic, and climatic evolution of the northern margin of the Tibetan Plateau. *Earth and Planetary Science Letters*, 282(1–4), 158–166. https://doi.org/10.1016/j.epsl.2009.03.011
- Klemperer, S. L. (2006). Crustal flow in Tibet: Geophysical evidence for the physical state of Tibetan lithosphere, and inferred patterns of active flow. In *Geological society special publication* (Vol. 268, pp. 39–70). https://doi.org/10.1144/GSL.SP.2006.268.01.03
- Lee, J., Hager, C., Wallis, S. R., Stockli, D. F., Whitehouse, M. J., Aoya, M., & Wang, Y. (2011). Middle to late Miocene extremely rapid exhumation and thermal reequilibration in the Kung Co rift, southern Tibet. *Tectonics*, 30, TC2007. https://doi.org/10.1029/2010TC002745
- Li, B., Zuza, A. V., Chen, X., Hu, D., Shao, Z., Qi, B., et al. (2020). Cenozoic multi-phase deformation in the Qilian Shan and out-of-sequence development of the northern Tibetan Plateau. *Tectonophysics*, 782–783, 228423. https://doi.org/10.1016/j.tecto.2020.228423
- Li, C., van der Hilst, R. D., & Toksöz, M. N. (2006). Constraining P-wave velocity variations in the upper mantle beneath Southeast Asia. *Physics of the Earth and Planetary Interiors*, 154, 180–195. https://doi.org/10.1016/j.pepi.2005.09.008
- Li, S., Currie, B. S., Rowley, D. B., & Ingalls, M. (2015). Cenozoic paleoaltimetry of the SE margin of the Tibetan Plateau: Constraints on the tectonic evolution of the region. *Earth and Planetary Science Letters*, 432, 415–424. https://doi.org/10.1016/j.epsl.2015.09.044
- Li, Y., Wang, C., Dai, J., Xu, G., Hou, Y., & Li, X. (2015). Propagation of the deformation and growth of the Tibetan-Himalayan orogen: A review. *Earth-Science Reviews*, 143, 36–61. https://doi.org/10.1016/j.earscirev.2015.01.001
- Liang, S., Gan, W., Shen, C., Xiao, G., Liu, J., Chen, W., et al. (2013). Three-dimensional velocity field of present-day crustal motion of the Tibetan Plateau derived from GPS measurements. *Journal of Geophysical Research: Solid Earth*, 118, 5722–5732. https://doi.org/10.1002/2013JB010503
- Liu, M., & Yang, Y. (2003). Extensional collapse of the Tibetan Plateau: Results of three-dimensional finite element modeling. *Journal of Geophysical Research*, 108, 2361. https://doi.org/10.1029/2002jb002248
- Maiti, G., & Mandal, N. (2021). Early Miocene exhumation of high-pressure rocks in the Himalaya: A response to reduced India-Asia convergence velocity. Frontiers in Earth Science, 9. https://doi.org/10.3389/feart.2021.632806
- Maiti, G., Mandal, N., & Misra, S. (2020). Insights into the dynamics of an orogenic wedge from lubrication theory: Implications for the Himalayan tectonics. *Tectonophysics*, 776, 228335. https://doi.org/10.1016/j.tecto.2020.228335
- Marques, F. O., & Mandal, N. (2016). Post-buckling relaxation of an elastic layer and its geological relevance: Insights from analogue experiments in pure shear. *Tectonophysics*, 668–669, 82–91. https://doi.org/10.1016/j.tecto.2015.12.004
- McCallister, A. T., Taylor, M. H., Murphy, M. A., Styron, R. H., & Stockli, D. F. (2014). Thermochronologic constraints on the late Cenozoic exhumation history of the Gurla Mandhata metamorphic core complex, Southwestern Tibet. *Tectonics*, 33, 27–52. https://doi.org/10.1002/2013TC003302

- Meigs, A. J., Burbank, D. W., & Beck, R. A. (1995). Middle-late Miocene (>10 Ma) formation of the Main Boundary Thrust in the western Himalaya. *Geology*, 23. https://doi.org/10.1130/0091-7613(1995)023<0423:MLMMFO>2.3.CO;2
- Meyer, B., Tapponnier, P., Bourjot, L., Métivier, F., Gaudemer, Y., Peltzer, G., et al. (1998). Crustal thickening in Gansu-Qinghai, lithospheric mantle subduction, and oblique, strike-slip controlled growth of the Tibet Plateau. *Geophysical Journal International*, *135*, 1–47. https://doi.org/10.1046/j.1365-246X.1998.00567.x
- Mitsuishi, M., Wallis, S. R., Aoya, M., Lee, J., & Wang, Y. (2012). E-W extension at 19Ma in the Kung Co area, S. Tibet: Evidence for contemporaneous E-W and N-S extension in the Himalayan orogen. *Earth and Planetary Science Letters*, 325–326, 10–20. https://doi. org/10.1016/j.epsl.2011.11.013
- Molnar, P., England, P., & Martinod, J. (1993). Mantle dynamics, uplift of the Tibetan Plateau, and the Indian Monsoon. Reviews of Geophysics, 31, 357. https://doi.org/10.1029/93RG02030
- Molnar, P., & Stock, J. M. (2009). Slowing of India's convergence with Eurasia since 20 Ma and its implications for Tibetan mantle dynamics. *Tectonics*, 28, TC3001. https://doi.org/10.1029/2008TC002271
- Molnar, P., & Tapponnier, P. (1975). Cenozoic Tectonics of Asia: Effects of a Continental Collision: Features of recent continental tectonics in Asia can be interpreted as results of the India-Eurasia collision. *Science*, *189*, 419–426. https://doi.org/10.1126/science.189.4201.419
- Montemagni, C., Iaccarino, S., Montomoli, C., Carosi, R., Jain, A. K., & Villa, I. M. (2018). Age constraints on the deformation style of the south Tibetan detachment system in Garhwal Himalaya. *Italian Journal of Geosciences*, 137, 175–187. https://doi.org/10.3301/ IJG.2018.07
- Murphy, M. A., Yin, A., Harrison, T. M., Dürr, S. B., Chen, Z., Ryerson, F. J., et al (1997). Did the Indo-Asian collision alone create the Tibetan plateau? *Geology*, 25. https://doi.org/10.1130/0091-7613(1997)025<0719:DTIACA>2.3.CO;2
- Najman, Y. (2006). The detrital record of orogenesis: A review of approaches and techniques used in the Himalayan sedimentary basins. *Earth-Science Reviews*, 74, 1–72. https://doi.org/10.1016/j.earscirev.2005.04.004
- Pan, Y., & Shen, W. B. (2017). Contemporary crustal movement of southeastern Tibet: Constraints from dense GPS measurements. Scientific Reports, 7. https://doi.org/10.1038/srep45348
- Penney, C., & Copley, A. (2021). Lateral variations in lower crustal strength control the temporal evolution of mountain ranges: Examples from south-east Tibet. *Geochemistry, Geophysics, Geosystems*, 22, e2020GC009092. https://doi.org/10.1029/2020GC009092
- Powell, C. M. A. (1986). Continental underplating model for the rise of the Tibetan Plateau. Earth and Planetary Science Letters, 81, 79–94. https://doi.org/10.1016/0012-821X(86)90102-0
- Pusok, A. E., & Kaus, B. J. P. (2015). Development of topography in 3-D continental-collision models. Geochemistry, Geophysics, Geosystems, 16, 1378–1400. https://doi.org/10.1002/2015GC005732
- Pusok, A. E., & Stegman, D. R. (2020). The convergence history of India-Eurasia records multiple subduction dynamics processes. Science Advances, 6. https://doi.org/10.1126/sciady.aa28681
- Regard, V., Faccenna, C., Martinod, J., & Bellier, O. (2005). Slab pull and indentation tectonics: Insights from 3D laboratory experiments. *Physics of the Earth and Planetary Interiors*, 149, 99–113. https://doi.org/10.1016/j.pepi.2004.08.011
- Rey, P., Vanderhaeghe, O., & Teyssier, C. (2001). Gravitational collapse of the continental crust: Definition, regimes and modes. *Tectonophysics*, 342, 435–449. https://doi.org/10.1016/S0040-1951(01)00174-3
- Ribe, N. M., & Davaille, A. (2013). Dynamical similarity and density (non-) proportionality in experimental tectonics. *Tectonophysics*, 608, 1371–1379. https://doi.org/10.1016/j.tecto.2013.06.005
- Rowley, D. B. (1998). Minimum age of initiation of collision between India and Asia North of Everest based on the subsidence history of the Zhepure Mountain section. *The Journal of Geology*, 106, 220–235. https://doi.org/10.1086/516018
- Royden, L. H., Burchfiel, B. C., King, R. W., Wang, E., Chen, Z., Shen, F., & Liu, Y. (1997). Surface deformation and lower crustal flow in eastern Tibet. Science, 276, 788–790. https://doi.org/10.1126/science.276.5313.788
- Royden, L. H., Burchfiel, B. C., & Van Der Hilst, R. D. (2008). The geological evolution of the Tibetan plateau. Science, 321, 1054–1058. https://doi.org/10.1126/science.1155371
- Rupnik, E., Daakir, M., & Pierrot Deseilligny, M. (2017). MicMac A free, open-source solution for photogrammetry. Open Geospatial Data, Software and Standards, 2. https://doi.org/10.1186/s40965-017-0027-2
- Schellart, W. P., Chen, Z., Strak, V., Duarte, J. C., & Rosas, F. M. (2019). Pacific subduction control on Asian continental deformation including Tibetan extension and eastward extrusion tectonics. *Nature Communications*, 10. https://doi.org/10.1038/s41467-019-12337-9
- Schellart, W. P., & Strak, V. (2016). A review of analogue modelling of geodynamic processes: Approaches, scaling, materials and quantification, with an application to subduction experiments. *Journal of Geodynamics*, 100, 7–32. https://doi.org/10.1016/j.jog.2016.03.009
- Song, B., Spicer, R. A., Zhang, K., Ji, J., Farnsworth, A., Hughes, A. C., et al. (2020). Qaidam Basin leaf fossils show northeastern Tibet was high, wet and cool in the early Oligocene. *Earth and Planetary Science Letters*, 537, 116175. https://doi.org/10.1016/j.epsl.2020.116175
- Soucy La Roche, R., Godin, L., Cottle, J. M., & Kellett, D. A. (2018). Preservation of the Early Evolution of the Himalayan Middle Crust in Foreland Klippen: Insights from the Karnali Klippe, West Nepal. *Tectonics*, *37*, 1161–1193. https://doi.org/10.1002/2017TC004847
- Spicer, R. A., Su, T., Valdes, P. J., Farnsworth, A., Wu, F. X., Shi, G., et al. (2021). Why "the uplift of the Tibetan Plateau" is a myth. *National Science Review*, 8. https://doi.org/10.1093/nsr/nwaa091
- Stevens, V. L., & Avouac, J. P. (2015). Interseismic coupling on the main Himalayan thrust. *Geophysical Research Letters*, 42(14), 5828–5837. https://doi.org/10.1002/2015GL064845
- Styron, R., Taylor, M., & Sundell, K. (2015). Accelerated extension of Tibet linked to the northward underthrusting of Indian crust. Nature Geoscience, 8, 131–134. https://doi.org/10.1038/ngeo2336
- Styron, R. H., Taylor, M. H., Sundell, K. E., Stockli, D. F., Oalmann, J. A. G., Möller, A., et al. (2013). Miocene initiation and acceleration of extension in the South Lunggar rift, western Tibet: Evolution of an active detachment system from structural mapping and (U-Th)/He thermochronology. *Tectonics*, 32, 880–907. https://doi.org/10.1002/tect.20053
- Tapponnier, P., Peltzer, G., Le Dain, A. Y., Armijo, R., & Cobbold, P. (1982). Propagating extrusion tectonics in Asia: New insights from simple experiments with plasticine. *Geology*, 10. https://doi.org/10.1130/0091-7613(1982)10<611:PETIAN>2.0.CO;2
- Tapponnier, P., Zhiqin, X., Roger, F., Meyer, B., Arnaud, N., Wittlinger, G., & Jingsui, Y. (2001). Oblique stepwise rise and growth of the Tibet plateau. *Science*, 294, 1671–1677. https://doi.org/10.1126/science.105978
- Taylor, M., Forte, A., Laskowski, A., & Ding, L. (2021). Active uplift of southern Tibet revealed. *Geological Society of America Today*, *31*(8), 4–10. https://doi.org/10.1130/GSATG487A.1
- Taylor, M., & Yin, A. (2009). Active structures of the Himalayan-Tibetan orogen and their relationships to earthquake distribution, contemporary strain field, and Cenozoic volcanism. *Geosphere*, 5, 199–214. https://doi.org/10.1130/GES00217.1
- Thielicke, W., & Stamhuis, E. (2010). Pivlab-time-resolved digital particle image velocimetry tool for matlab. ... under the BSD License, Programmed with MATLAB. https://doi.org/10.6084/m9.figshare.1092508

- Van Hinsbergen, D. J. J., Kapp, P., Dupont-Nivet, G., Lippert, P. C., Decelles, P. G., & Torsvik, T. H. (2011). Restoration of Cenozoic deformation in Asia and the size of Greater India. *Tectonics*, 30, TC5003. https://doi.org/10.1029/2011TC002908
- Wallis, S., Tsujimori, T., Aoya, M., Kawakami, T., Terada, K., Suzuki, K., & Hyodo, H. (2003). Cenozoic and Mesozoic metamorphism in the Longmenshan orogen: Implications for geodynamic models of eastern Tibet. *Geology*, 31(9), 745–748. https://doi.org/10.1130/G19562.1
- Wang, C., Dai, J., Zhao, X., Li, Y., Graham, S. A., He, D., et al. (2014). Outward-growth of the Tibetan Plateau during the Cenozoic: A review. Tectonophysics, 621, 1–43. https://doi.org/10.1016/j.tecto.2014.01.036
- Wang, C., Zhao, X., Liu, Z., Lippert, P. C., Graham, S. A., Coe, R. S., et al. (2008). Constraints on the early uplift history of the Tibetan Plateau. Proceedings of the National Academy of Sciences of the United States of America, 105, 4987–4992. https://doi.org/10.1073/ pnas.0703595105
- Wang, E., Kirby, E., Furlong, K. P., Van Soest, M., Xu, G., Shi, X., et al. (2012). Two-phase growth of high topography in eastern Tibet during the Cenozoic. Nature Geoscience, 5, 640–645. https://doi.org/10.1038/ngeo1538
- Wang, L., Cheng, F., Zuza, A. V., Jolivet, M., Liu, Y., Guo, Z., et al. (2021). Diachronous growth of the northern Tibetan Plateau derived from flexural modeling. *Geophysical Research Letters*, 48(8), 1–13. https://doi.org/10.1029/2020GL092346
- Wang, Q., Zhang, P. Z., Freymueller, J. T., Bilham, R., Larson, K. M., Lai, X., et al. (2001). Present-day crustal deformation in China constrained by global positioning system measurements. *Science*, 294, 574–577. https://doi.org/10.1126/science.1063647
- Wang, S., Chen, Y., Charreau, J., Li, Y., Chen, Z., Zhu, G., et al. (2021). Underthrusting of the Tarim lithosphere beneath the Western Kunlun Range, insights from seismic profiling evidence. *Tectonics*, 40(2), 1–18. https://doi.org/10.1029/2019TC005932
- Wang, Y. (2001). Heat flow pattern and lateral variations of lithosphere strength in China mainland: Constraints on active deformation. Physics of the Earth and Planetary Interiors, 126, 121–146. https://doi.org/10.1016/S0031-9201(01)00251-5
- Webb, A. A. G., Guo, H., Clift, P. D., Husson, L., Müller, T., Costantino, D., et al. (2017). The Himalaya in 3D: Slab dynamics controlled mountain building and monsoon intensification. *Lithosphere*, 9(4), 637–651. https://doi.org/10.1130/L636.1
- Wei, H. H., Meng, Q. R., Ding, L., & Li, Z. Y. (2013). Tertiary evolution of the western Tarim basin, northwest China: A tectono-sedimentary response to northward indentation of the Pamir salient. *Tectonics*, 32, 558–575. https://doi.org/10.1002/tect.20046
- Wilson, C. J. L., & Fowler, A. P. (2011). Denudational response to surface uplift in east Tibet: Evidence from apatite fission-track thermochronology. Bulletin of the Geological Society of America, 123, 1966–1987. https://doi.org/10.1130/B30331.1
- Woodruff, W. H., Horton, B. K., Kapp, P., & Stockli, D. F. (2013). Late Cenozoic evolution of the Lunggar extensional basin, Tibet: Implications for basin growth and exhumation in hinterland plateaus. *Bulletin of the Geological Society of America*, 125, 343–358. https://doi. org/10.1130/B30664.1
- Wu, C., Li, J., Zuza, A. V., Liu, C., Liu, W., Chen, X., et al. (2020). Cenozoic cooling history and fluvial terrace development of the western domain of the Eastern Kunlun Range, northern Tibet. Palaeogeography, Palaeoclimatology, Palaeoecology, 560, 109971. https://doi. org/10.1016/j.palaeo.2020.109971
- Wu, C., Zuza, A. V., Li, J., Haproff, P. J., Yin, A., Chen, X., et al. (2021). Late Mesozoic–Cenozoic cooling history of the northeastern Tibetan Plateau and its foreland derived from low-temperature thermochronology. GSA Bulletin, 133, 2393–2417. https://doi.org/10.1130/ b35879.1
- Xu, G., & Kamp, P. J. J. (2000). Tectonics and denudation adjacent to the Xianshuihe Fault, eastern Tibetan Plateau: Constraints from fission track thermochronology. Journal of Geophysical Research, 105, 19231–19251. https://doi.org/10.1029/2000jb900159
- Yakovlev, P. V. (2015). Evolution of the Indo-Asian orogen: Insights from the deformation of the northern Tibetan plateau, mass balance calculations, and volcanic geochemistry. Ph.D. Thesis. Michigan University. Retrieved from https://deepblue.lib.umich.edu/ handle/2027.42/113409
- Yang, S., Chen, H., & Cheng, X. (2007). Deformation characteristics and rules of spatial change for the Northern Qilianshan thrust belt. Earth Science Frontiers, 14(5), 211–221. https://doi.org/10.3321/j.issn:1005-2321.2007.05.021
- Yang, Y., & Liu, M. (2013). The indo-Asian continental collision: A 3-D viscous model. *Tectonophysics*, 606, 198–211. https://doi. org/10.1016/j.tecto.2013.06.032
- Yin, A. (2000). Mode of Cenozoic east-west extension in Tibet suggesting a common origin of rifts in Asia during the Indo-Asian collision. Journal of Geophysical Research, 105, 21745–21759. https://doi.org/10.1029/2000jb900168
- Yin, A., Dang, Y. Q., Wang, L. C., Jiang, W. M., Zhou, S. P., Chen, X. H., et al. (2008). Cenozoic tectonic evolution of Qaidam basin and its surrounding regions (Part 1): The southern Qilian Shan-Nan Shan thrust belt and northern Qaidam basin. Bulletin of the Geological Society of America, 120, 813–846. https://doi.org/10.1130/B26180.1
- Yin, A., Rumelhart, P. E., Butler, R., Cowgill, E., Harrison, T. M., Foster, D. A., et al (2002). Tectonic history of the Altyn Tagh fault system in northern Tibet inferred from Cenozoic sedimentation. *Bulletin of the Geological Society of America*, 114, 1257–1295. https://doi.org/1 0.1130/0016-7606(2002)114<1257:THOTAT>2.0.CO;2
- Yin, A., & Taylor, M. H. (2011). Mechanics of V-shaped conjugate strike-slip faults and the corresponding continuum mode of continental deformation. *Bulletin of the Geological Society of America*, 123, 1798–1821. https://doi.org/10.1130/B30159.1
- Zhang, P. Z., Shen, Z., Wang, M., Gan, W., Bürgmann, R., Molnar, P., et al. (2004). Continuous deformation of the Tibetan Plateau from global positioning system data. *Geology*, *32*, 809. https://doi.org/10.1130/G20554.1
- Zhao, W.-L., & Morgan, W. J. (1987). Injection of Indian crust into Tibetan lower crust: A two-dimensional finite element model study. *Tectonics*, 6, 489–504. https://doi.org/10.1029/TC006i004p00489
- Zheng, G., Wang, H., Wright, T. J., Lou, Y., Zhang, R., Zhang, W., et al. (2017). Crustal deformation in the India-Eurasia collision zone from 25 years of gps measurements. *Journal of Geophysical Research: Solid Earth*, 122, 9290–9312. https://doi.org/10.1002/2017JB014465
- Zulauf, J., & Zulauf, G. (2004). Rheology of plasticine used as rock analogue: The impact of temperature, composition and strain. *Journal of Structural Geology*, *26*, 725–737. https://doi.org/10.1016/j.jsg.2003.07.005
- Zuza, A. V., & Yin, A. (2017). Balkatach hypothesis: A new model for the evolution of the Pacific, Tethyan, and Paleo-Asian oceanic domains. *Geosphere*, 13, 1664–1712. https://doi.org/10.1130/GES01463.1